

# Interviewing Child Witnesses under the Memorandum of Good Practice: A research review

Graham M. Davies Helen L. Westcott

Police Research Series Paper 115

# Interviewing Child Witnesses under the *Memorandum of Good Practice*: A research review

Graham M. Davies Helen L. Westcott



Editor: Barry Webb Home Office Policing and Reducing Crime Unit Research, Development and Statistics Directorate Clive House, Petty France London, SW1H 9HD © Crown Copyright 1999 First Published 1999

# Policing and Reducing Crime Unit: Police Research Series

The Policing and Reducing Crime Unit (PRC) was formed in 1998 as a result of the merger of the Police Research Group (PRG) and the Research and Statistics Directorate. PRC is now part of the Research, Development and Statistics Directorate of the Home Office. PRC carries out and commissions research in the social and management sciences on policing and crime reduction.

PRC has now combined PRG's two main series into the Police Research Series. The series will present research material on crime prevention and detection as well as police management and organisation issues.

Research commissioned by PRG will appear as a PRC publication.

ISBN 1-84082-328-3

Copies of this publication can be made available in formats accessible to the visually impaired on request.

# **Foreword**

The 'Memorandum of Good Practice on Video Recorded Interviews with Child Witnesses for Criminal Proceedings' (Home Office/Department of Health) was published in 1992 to provide guidance to police officers and social workers responsible for undertaking video-recorded interviews with child victims or witnesses. The document outlined core principles to be followed when conducting interviews; the video could then be played in court to spare the child the necessity of giving live examination-in-chief.

Since 1992, much literature and research has been generated around the *Memorandum* and around the issue of child witnesses in general. This report reviews this literature, drawing out the implications that this has for interviews conducted under *Memorandum* guidelines. It will therefore be of relevance for police officers, social workers and other agencies involved in investigations and interviews with children.

The report also complements recent policy recommendations, such as those contained in the Home Office consultation document 'Speaking Up For Justice', some of which are included in the Youth Justice and Criminal Evidence Bill. It should also help to inform the revision of the original Memorandum, which is being taken forward as part of the programme of implementation of 'Speaking Up For Justice'.

Gloria Laycock Policing and Reducing Crime Unit Home Office September 1999

# Acknowledgements

The authors wish to thank Sally Pearson for her valuable research assistance, Julie Taylor-Browne for her encouragement and patience, and Emma Marshall for her work in seeing this report through to publication.

## The Authors

- Graham Davies is a Professor of psychology at Leicester University; and,
- Helen Westcott is a lecturer in psychology at The Open University.

PRCU would like to thank Professor Ray Bull, Portsmouth University, who acted as external assessor for this report.

# **Executive summary**

This report summarises the findings of recent research on children as witnesses and draws out the implications for the conduct of interviews under the 'Memorandum of Good Practice on Video Recorded Interviews with Child Witnesses for Criminal Proceedings' (Home Office/Department of Health, 1992). Most of the research reviewed here has been conducted or published since the Memorandum was drafted. The report does not, however, pretend to be an exhaustive academic literature review, but aims to focus on the practical relevance that studies in psychology, social work and policing have for investigative interviewing.

# Child development

Research on children's developing competence as witnesses highlights:

- the vulnerabilities of very young witnesses;
- the value of prompt interviewing;
- ways of reducing the possibility that children will go along with suggestive questions;
- the importance of accommodating to the child's language and level of understanding;
- the importance of asking questions at a level, and in a form, appropriate to the individual child;
- whether credibility can be assessed through the content of the child's statement, or their non-verbal behaviour; and,
- the importance of listening to what the child has to say.

#### Planning the interview

When planning an investigative interview, interviewers need to be aware of:

- differences in the way children disclose abuse;
- proper preparation of the child for the interview, and giving the child as much choice (and thus control) as possible;
- tailoring the interview to take account of factors such as the child's age, state of mind and level of anxiety, physical or learning impairments, and race and culture; and.
- the influence of the interviewer; children need support throughout the interview.

#### Conducting the interview

This section reviews research and practice in conducting *Memorandum* interviews. It indicates that:

 the rapport phase of the interview can be used not merely to reduce social distance, but also to determine children's level of understanding and

- competence and provide an opportunity to lay appropriate ground rules for the interview;
- although open questions provide more accurate information than closed questions, in some circumstances direct prompts are needed. In these situations, inappropriate questioning techniques, such as forced choice questions, multipart questions or 'can you...' questions should be avoided; and,
- proper closure of the interview is frequently neglected; it should provide an opportunity for the interviewers to answer any questions children may have and to thank them for their time and effort.

The report also suggests that further guidance is needed on:

- tests of 'truth/lies';
- the duration, pace and number of interviews;
- the potential value of drawings, props, toys and anatomically detailed dolls; and,
- specialised interview techniques, such as the Cognitive Interview, semi-scripted interviews and SAGE (Systematic Approach to Gathering Evidence).

# Appearance at court

The literature on child witness preparation programmes and the long-term effects of appearing in court on children, concludes that:

- the impact of a court appearance can be exacerbated by poorly conducted repeated interviews, harsh cross-examination or by testifying more than once;
- maternal support, case resolution and the passage of time ameliorate the effects of court appearance;
- preparation programmes are beneficial to children, but are under-funded and not sufficiently evaluated; and,
- experimental research on preparation suggests some potentially valuable practical interventions which deserve to be tried out in realistic settings.

The final section of the report reviews the main themes and recommendations for any future revision of the *Memorandum*. Recent policy developments such as the Utting report (1997) and 'Speaking Up for Justice' (1998) suggest that changes in the criminal justice system as it affects child witnesses will continue, and this is reflected in the 1999 Youth Justice and Criminal Evidence Bill. This report urges that any revision to the Memorandum is informed by the research reviewed here, and is located within a strategic, comprehensive and nationally agreed framework of reform.

# Contents

	Page
Foreword	(iii)
Acknowledgements	(iv)
Executive summary	(v)
1. Introduction  Aims and scope of the report  The 'Memorandum of Good Practice'  Child witness research  Structure of the report	1 1 1 6 6
2. Child development  Memory Suggestibility The type of language used in the interview Deception	8 8 9 10 12
3. Planning the interview Differences in childrens' disclosure of abuse Preparing the child for the interview Characteristics of the child and family The influence of the interviewer	14 14 15 15
4. Conducting the <i>Memorandum</i> interview The phased approach Duration, pace and number of interviews Assisting child witnesses in the interview Alternative questioning techniques	20 20 23 24 26
<b>5.</b> Appearance at court Long-term effects of court appearance Preparation of child witnesses for court	28 28 28
6. Review Child development Planning the interview Conducting the Memorandum interview Appearance at court Concluding comments	32 32 33 35 37 38
References	40
Recent PRCU publications	57

# 1. Introduction

# Aims and scope of the report

Children involved in legal proceedings concerning child abuse may have to tell their story on several occasions. This starts with the initial interview and may culminate in examination and cross-examination at court. Where there is a possibility that the child's statement may be used in evidence they may be video-interviewed by a police officer or social worker. These interviews are conducted under the guidelines set out by the 'Memorandum of Good Practice on Video Recorded Interviews with Child Witnesses for Criminal Proceedings' (Home Office/Department of Health, 1992: hereafter the Memorandum). This report summarises the implications that recent research on children as witnesses has for the conduct of interviews under the Memorandum.

Most of the research reviewed here has been conducted since the Memorandum was issued in 1992. While the report sometimes cites the practice and research of knowledgeable professionals, wherever possible the major recommendations are based upon the results of experimental research, courtroom observation or analysis of actual investigative interviews. It does not claim to be an exhaustive review of the literature, but instead focuses on the practical relevance of studies from psychology, social work and policing.

## 'The Memorandum of Good Practice'

The origins of the Memorandum

The origins of the *Memorandum* lie in the report of an advisory group, chaired by His Honour Judge Thomas Pigot QC, which had been set up to consider the admissibility of video-recorded interviews with children in criminal cases. The Pigot Report (1989) recommended that such interviews, conducted by a police officer or social worker, should be used as a substitute for the child's live examination-in-chief at trial.

To ensure that the interviews were carried out in accord with the rules of evidence, the report recommended that a 'Code of Practice' be drawn up to govern their conduct. It also recommended that the videotape principle should be extended to examination by the defence, so that tapes of these interviews could replace the child's cross-examination at trial. In addition, the use of intermediaries was proposed to assist examination of young or otherwise vulnerable children at trial. These measures could spare many children the necessity of attending a formal trial. However, although the Criminal Justice Act 1991 incorporated the proposals regarding the admissibility of videotaped interviews as evidence-in-chief, it did not follow the report's recommendations regarding cross-examination and intermediaries.

1

In compiling a draft Code of Practice, the Home Office commissioned a psychologist, Professor Ray Bull, and a lawyer, Dr (now Professor) Diane Birch. The subsequent draft underwent successive revisions by a working party of concerned professionals (under the auspices of Home Office officials). During this process, the title changed from 'Code' to 'Memorandum of Good Practice' to reflect the view that the document should provide guidance rather than seek to lay down inflexible rules. The Memorandum was launched in August 1992 to coincide with the implementation of the 1991 Act.

The main recommendations of the Memorandum are that:

- interviews should be conducted as soon as practicable after an allegation of abuse emerges;
- interviews should take place in an informal setting with interviewers trained in talking to children;
- children should be given every opportunity to tell their own story before being asked explicit questions;
- questioning should follow a phased or 'step-wise' approach, beginning with open-ended queries and reserving any direct or leading questions for the final phase of the interview; and,
- as a general rule, interviews should last no longer than one hour.

# The 'Memorandum' in practice

Shortly after the implementation of the 1991 Act, a survey was conducted by the Association of Chief Police Officers (Butler, 1993). This revealed that nearly 15,000 interviews had been conducted under the Act in its first nine months of operation. However, less than a quarter of the videotaped interviews had been submitted to the CPS for prosecution purposes, and only 44 were known to have been played at court. At face value, these figures painted a pessimistic picture of the effectiveness of the new procedures. However, legal delays probably underestimated the number of video cases which eventually reached court, and the figures omitted cases where late guilty pleas removed the need to show the videotaped interview. The same survey also highlighted widespread disparities between forces in rates of interviewing and submission of cases to the CPS.

The Home Office also commissioned an initial evaluation of the new procedures. The subsequent report (Davies et al., 1995) concluded that:

- within 18 months of implementation of the Act, 75% of relevant cases included an application to show a videotaped interview at trial;
- there was widespread acceptance among police officers and social workers of the evidential value of videotaped interviews and of the *Memorandum*;

- judges generally expressed positive attitudes toward the new legislation, but barristers showed less enthusiasm;
- a sample of taped interviews showed that interviewers secured a clear account of events from the child in 75% of cases;
- children who gave their evidence on tape were more relaxed than those testifying live at court; and,
- interviewers did not always follow the *Memorandum*'s emphasis on free narrative and open-ended questions.

The report also noted some specific criticisms of the Memorandum. In particular:

- that it had an emphasis on evidence gathering rather than support for the child:
- it gave inadequate guidance on interviews with young children or those with special needs; and,
- there were too few examples of good practice.

The study also noted a demand for consistent training standards, a view supported by a parallel study conducted by the Social Services Inspectorate (SSI; Holton and Bonnerjea, 1994).

The SSI study identified other perceived weaknesses in the Memorandum including:

- there was insufficient guidance on when to interview;
- the 'one hour' recommendation on interview length was not realistic;
- a neutral interviewing style was incompatible with proper support for the child victim;
- insufficient attention was given to the needs and wishes of the child; and,
- there was inadequate coverage of child development issues.

Subsequent research has continued to highlight criticisms, both of the content of the *Memorandum*, and of its interpretation by the courts. Indeed, a number of police child protection officers interviewed by Hughes, Gallagher and Parker (1996) doubted the value of the *Memorandum* and called for the document to be rewritten. However, a second study (Davies, et al., 1998) found continuing support for the principles of the document among police officers at all levels within child protection units, although they did reiterate many of the detailed criticisms mentioned above. In addition, officers expressed concern that the courts treated the *Memorandum* as an inflexible code rather than as guidance, and that cases were being dismissed because of what officers felt were relatively minor deviations from it. The research also found continuing support for national training guidelines (Davies et al., 1998).

Other criticisms have included the *Memorandum*'s failure to address issues concerning children with special needs. Westcott and Jones (1997) highlighted the difficulties faced by disabled and black children, and concluded that the gulf between the needs of the abused child and the demands of the criminal justice system were likely to remain unbridgeable within current policy and practice. They called for a review of alternative systems of evidence gathering, such as forensic analysis and the development of interviewing strategies for suspected abusers.

Whilst most of the principles of the *Memorandum* apparently enjoy widespread support, other aspects appear to require revision and reflection. A review of its content is timely, given recent official reports which have highlighted the dilemmas facing child witnesses under existing law (Utting et al., 1997; 'Speaking up for Justice', 1998; Davis et al., in press). A number of reforms, including full implementation of the Pigot proposals, are incorporated into the Youth Justice and Criminal Evidence Bill, currently before Parliament. It is important to remember, however, that this latter Bill, like the Memorandum, applies only to England and Wales. Although much of this review may be relevant to practice in Scotland and/or Northern Ireland, significant differences in the investigative and criminal justice process exist.

#### Profile of cases and witnesses

What is known of the typical witness interviewed under the *Memorandum*? Figures supplied to Davies et al. (1995) by the Lord Chancellor's Department, covering 1,561 child witnesses giving evidence at court between October 1994 and April 1995, gave the following profile:

- 88% were alleged victims;
- 73% were female:
- the majority were aged between 10 and 15 years of age; and,
- 59% of cases involved a single child witness.

The study found few instances of children from minority ethnic communities or disabled children giving evidence in court. Later research also confirms that it is unusual for children with learning difficulties (Saunders et al., 1997) or those with other communication difficulties to be given a *Memorandum* interview, largely because of its emphasis upon the need for free narrative (Westcott, 1994). Although data are not available on the proportion of children from minority ethnic communities who participate in *Memorandum* interviews, a survey conducted for Victim Support suggested that approximately 8.5% of all child witnesses were from minority communities (Chandler and Lait, 1996). This is slightly less than the recorded population for persons under the age of 16 years from the ethnic

population (9.7%; Office for National Statistics, 1991). However, such comparisons are inevitably tentative and more comprehensive monitoring, both of victimisation and involvement with the criminal justice system, is required.

Regarding the alleged offence, Davies et al. (1995) reported that:

- 97% of all charges involved a single defendant;
- 96% of defendants were male;
- 76% of defendants were charged with indecent assault; and,
- most allegations concerned sexual, not physical assault.

The impact of Memorandum interviews on the prosecution process

Since Butler's (1993) study, no national surveys have examined the frequency of *Memorandum* interviews or their impact on the prosecution process. The lack of systematic data collection on prosecution and conviction rates in relation to child witnesses makes it difficult to evaluate legal and procedural reforms. The recent CPS Inspectorate Report (1998) does, however, confirm that the submission of an application to show a videotaped interview is now the norm in cases of alleged child sexual abuse. The inspectors reported that:

- 93% of all cases were accompanied by an application to show a video;
- 77% were granted (6% were out of time);
- 54% of videos required some editing before being shown to the court;
- 23% of applications granted to show a video were not taken up; and,
- 56% of children had their videotapes played at court.

Some of this attrition can be explained by late guilty pleas, but others arose from a change in tactics by prosecuting counsel in order to examine the child live at court, in the belief that their evidence would have more impact (Chandler and Lait, 1996; Davies et al., 1995). However, Davies et al. (1995) could find no evidence from successful prosecution figures to support this assumption, and noted that such tactical switches were usually unsuccessful and frequently had an adverse effect upon the child.

The Criminal Procedure and Investigations Act (1996) proposed that binding Plea and Directions Hearings into child witness cases should be introduced in part to ensure that when a decision was made to show a videotaped interview, it should only be reversed in exceptional circumstances. This section of the legislation was not implemented, but similar provisions are incorporated in the *Youth Justice and Criminal Evidence Bill* currently before Parliament. Finally, whilst the CPS report noted that over 17% of child complainants had their cases heard at magistrates' courts, which are not equipped with video facilities, the current Home Office

consultative paper on victims (Home Office, 1998) envisages an extension of such facilities to the lower courts (see Davis et al., in press, for a detailed discussion of the legal problems of prosecuting defendants on the basis of children's evidence).

#### Child witness research

Two types of research study are drawn on for this report:

- Field studies which have examined the process of gaining and giving evidence by child witnesses in England and Wales. These include courtroom observation studies and evaluation of video-recorded interviews; and,
- Experimental studies which seek to simulate, under laboratory conditions, the pressures and circumstances child witnesses encounter in observing and recounting experiences. These studies typically involve non-abused school children as participants and thus doubts inevitably arise as to the generalisability and representativeness of their findings. These experiments do, however, enable researchers to study a single issue with a precision rarely possible in field studies (Davies, 1992).

Most experimental studies originate in the United States and tend to reflect American concerns. For example, there are many studies on memory and suggestibility in children aged four to six years. These reflect a series of high profile, controversial cases involving adults convicted of sexual abuse on the testimony of pre-school children (Ceci and Bruck, 1995). In England and Wales, however, it is rare for children of this age to feature in court proceedings (Barker et al., 1998). The US legal system also permits the multiple interviewing of individual children and this in turn leads to considerable research on the cumulative impact of such interviews. By contrast, much research in the UK has focused on the effectiveness of specialist interviewing procedures or innovative questioning techniques (Memon et al., 1998).

# Structure of the report

The remainder of this report is structured as follows:

- The next section examines issues of child development. Recent experimental studies on children's memory, suggestibility, language development and deception are reviewed, and their implications for interviewing children of different ages are discussed;
- Section 3 considers planning the interview, taking into consideration differences in childrens' disclosure of abuse, preparing the child,

characteristics of the child and family background, and the influence of the interviewer;

- Section 4 covers the phases of the interview and draws on reports which have evaluated the content of actual interviews. The section also considers progress on innovative interviewing techniques and the use of props and cues;
- Section 5 describes studies on preparing children for their court appearance and the long-term consequences of giving evidence at court; and,
- Section 6 discusses the implications of research for police practice and summarises the main recommendations of this report.

# 2. Child development

Since the publication of the *Memorandum*, research has tried to identify how childrens' competence as witnesses grows with age. The findings of such research are of particular relevance to those who interview children involved in abuse allegations, and have been recently highlighted as important for police training (see Davies et al., 1998).

Four major areas relevant to this issue are:

- memory;
- suggestibility;
- language; and,
- deception.

#### Memory

The *Memorandum* interview currently consists of phases that combine elements of free recall and prompted recall. Researchers therefore distinguish between children's unprompted statements and information arising from questions.

Free recall

With regard to free recall within interviews, the consensus among researchers is that:

- the quantity of children's free recall increases with age;
- free recall is generally very accurate;
- the accuracy of reports do not vary with age; and,
- the omission of details is much more common than the invention of false ones.

This has been supported by studies involving a range of witnessed events, including commonly occurring stressful incidents, such as medical interventions or dental visits (Ornstein et al., 1997; Goodman and Schaaf, 1997). Although no clear relationship has emerged between the rated stressfulness of such incidents for the child and amount recalled (Vandermaas et al., 1993; Merritt et al., 1994), it is clear that the amount and accuracy of recall deteriorates over time (Flin et al., 1992; Poole and White, 1993). In addition, younger children (three to six years) appear to forget more rapidly than older children or adults (Baker-Ward et al., 1993; Poole and White, 1993). However, although their free recall is typically more incomplete and brief compared to older children, it is no less accurate (Saywitz et al., 1996).

Although children as young as three to four years are capable of spontaneous and accurate recall of events from their own lives, this may be limited to one or two salient facts (Fundudis, 1997). In addition, while isolated examples may be found of

young children introducing fantastical elements into their reports, this is exceptional and does not necessarily invalidate other parts of the child's statement (Everson, 1997).

#### Prompted recall

All witnesses know more than they are able to spontaneously recall and questioning is required to assist retrieval of relevant information from memory. However, experimental studies suggest that:

- although questioning increases the amount of information provided, prompted recall is less accurate than free recall;
- open-ended questions are answered more accurately than specific questions;
- specific questions are answered more accurately than leading questions; and,
- loss of accuracy is greater in younger than in older children.

The Memorandum correctly emphasises the evidential value of responses to openended questions and discourages the use of leading questions.

#### Suggestibility

Suggestibility has been defined as 'the act or process of impressing something (an idea, attitude or desired action) on the mind of another' (Fundudis, 1997: 151). Much has been learned recently about the circumstances in which children are likely to go along with the implications of leading questions. The idea that children are infinitely suggestible and can be encouraged to make plausible allegations of abuse against an adult on the basis of a few leading questions has been refuted by research (Goodman et al., 1991). Nonetheless, there are circumstances under which children can be vulnerable to both cognitive and social forms of suggestion, i.e:

- the interrogation process has distorted their memory; and/or,
- they are acquiescing to the views of the more powerful interviewer.

An influential review of experimental research by Ceci and Bruck (1993) concluded that children in general, and young children below six years in particular, were prone to incorrect responding under certain circumstances (see McAuliff et al., 1998 for an update and critique), including:

An accusatory context. In several experiments, neutral or ambiguous
actions were performed by an adult. These were later reinterpreted
negatively by young children if the interviewer or another authority figure
repeatedly suggested that the adult's behaviour was suspicious (Leichtman
and Ceci, 1995; Lepore and Sesco, 1994; Thompson et al., 1997). Gently

challenging children about such beliefs can substantially reduce, but not eliminate, this problem.

- Repeated suggestive interviewing . Both adult and child witnesses have been shown to be susceptible to leading questions. However, children below six appear especially vulnerable (see Cassel and Bjorklund, 1995). Repeated interviews with four to six year olds, in which leading questions were used to imply the same misleading account of events, led to free recall as well as prompted recall being affected, with convincing but false corroborating detail being offered by the child (Leichtman and Ceci, 1995).
- Post-event misinformation. Both adult and child witnesses who observe an
  event and later read a misleading account of it will incorporate some of the
  misinformation into their memories of the original event. Once again,
  however, four to six year olds appear particularly vulnerable to this form of
  distortion, which influences free as well as prompted recall (Poole and
  Lindsay, 1995).
- Memories implanted by others. It appears to be much easier to change a memory than to implant a totally false one in a witness' memory (Pezdek and Roe, 1997). However, adults repeatedly presented with a mix of real and fictitious events from their own lives will over time claim to 'remember' at least some of the fictitious events (Hyman et al., 1995). Similar effects have been demonstrated in four to six year old children, some of whom claimed not only that the fictitious events had occurred, but also provided plausible supportive detail. This effect was maximised if witnesses were encouraged to visualise repeatedly each incident and were given an assurance that their parents remembered all the events (Ceci et al., 1994). However, for such effects to occur, the suggested event must be compatible with the child's previous experience and beliefs (Pezdek et al., 1998).

# The type of language used in the interview

Much of the research on the use of language in interviewing has focused on the courtroom. This has highlighted the problems which can arise when language inappropriate to the age of the child is used (Westcott, 1995; Wilson, 1995). The use of adult language in questioning a child can lead to inconsistent and confused answers and to an increase in incorrect responding (Carter et al., 1996). Children appear reluctant to query questions they do not understand, sometimes even when the question is not sensible.

Davies et al. (1995) found that although *Memorandum* interviewers tailor their language to the age of the child rather better than court officials, there can still be short-comings in the interviews (Westcott and Davies, 1996b). Bull (1995) and others found that not enough interviewers use the rapport phase of the interview to establish the child's level of linguistic competence. Three other areas of concern have also been identified:

- Styles of questioning. Walker (1993; 1994) has suggested that the following range of questioning styles can cause problems for children, and therefore should be avoided by interviewers:
  - passive questions (e.g. 'were you chased by him?');
  - negative questions (e.g. 'did you not see him in the park?'); and,
  - questions containing multiple propositions (e.g. 'Did you go to your uncle's house and was he present when you took a bath?').

Walker suggests that younger children have particular problems in answering 'yes/no' questions accurately and respond better to 'wh-' questions (who, whom, what, where, when?).

- Vocabulary, particularly in relation to legal and sexual terms. Major confusions can arise from vocabulary which is not tailored to the age of the child. Simply asking the child whether a term is understood is insufficient as the child may use the term in a limited or idiosyncratic way. For example, the word 'jury' may mean 'something you wear around your neck' to a younger child (Warren and McGough, 1996).
- Comprehension of adult concepts. Many concepts which are taken for granted in adult conversation are only acquired gradually in childhood. Therefore, the introduction of these concepts may generate a misleading response and a subsequent loss of credibility in other aspects of the child's statement. Difficult concepts include:
  - date and time:
  - duration:
  - frequency;
  - measurement; and,
  - location.

The concept of time is only slowly mastered: the ability to tell the time is normally learned around six to seven years of age, while understanding the day of the week and the seasons is generally acquired around eight years. However, this does not automatically bestow accuracy in either locating events in time or estimating their

duration (Saywitz et al.,1993a). Children below eight years will have difficulty in answering accurately whether a given event in the past occurred before or after another or in estimating its duration (Saywitz, 1995). Saywitz et al. (1991) have demonstrated, however, that reference to events in the child's own life, such as television programmes, can provide an accurate indirect measure.

Similarly, the ability to count is essential, but does not in itself guarantee accuracy in estimating the frequency of an event. The more concrete terms used to frame the question, the greater the likelihood of an accurate response (Saywitz et al., 1993a). Accurate estimates of height, weight and age are notoriously difficult, even for adults, and children have particular problems with such measures. The use of relative judgements (e.g. 'was he shorter/taller than me?') provides more realistic estimates (Saywitz and Elliott, in press). Location is another area of difficulty: young children do not understand terms like 'behind', 'in front of', 'beneath' and 'above' (Wilson, 1995). However, statements regarding what can be expected of children of a given age must bear in mind that individual witnesses may perform above or below the level of their age group.

#### Deception

Three sources of information are available to interviewers as cues to deception:

Non-verbal behaviour and speech disturbances

Considerable research has been devoted to the use of non-verbal and speech cues in detecting deception in adults (Memon et al., 1998). The research concludes that:

- some cues do vary between deceptive and truthful statements;
- no single cue is uniquely associated with deception;
- cues linked to stress can be confused with those linked to lying; and,
- the public and professionals involved in child protection have an uncertain grasp of what the useful cues are.

#### Statement content

The analysis of statement content was a technique pioneered as an alternative to detecting truthfulness from children's demeanour (Bradford, 1994). Criteria-Based Content Analysis (CBCA) specifies 19 criteria, whose presence in a witness account is believed to characterise truthfulness (Raskin and Esplin, 1991). Comparisons of the content of interviews from children making well-founded and suspect allegations of abuse have suggested that at least some of the criteria have validity, though the differences are often small (Lamb et al., 1997). There are, however, continuing concerns over the reliability of scoring some criteria

(Horowitz et al., in press) and indications that considerable expertise is required to use the system effectively (Yuille, personal communication). CBCA may also be unsuitable for children who provide limited narratives, because of stress, depression or limited language development (Bekerian and Dennett, 1992).

Research reviewed in this section has highlighted the need for training and practice to emphasise:

- the value of free recall in interviews, and the need to minimise the number of closed and leading questions put to children;
- that interviewers should be aware of the dangers of children going along with implications of suggestive questioning in interviews, and the circumstances which can give rise to this;
- that the language used in interviews should be appropriate to age of the child;
- the provision of concrete examples to assist the child in understanding some concepts, e.g. dates, times and locations; and,
- although non-verbal cues and analysis of statement content can be used to detect deception, concerns currently exist over the reliability of these in practice.

# 3. Planning the interview

When planning an interview, a range of issues, other than those purely concerned with the interview, need to be taken into account. This will maximise the chances of the interview being successful. This section discusses the following:

- differences in childrens' disclosure of abuse;
- issues to consider when preparing the child for the interview;
- relevant characteristics of the child and family background; and,
- the influence of the interviewer on testimony.

## Differences in childrens' disclosure of abuse

Reviews of the *Memorandum* have argued that the existing guidance does not sufficiently describe the conditions, situations and children for whom *Memorandum* interviewing should be used. The videotaping of every investigative interview with a child is neither resource-sustainable nor desirable, and local practices mean that the number of video interviews vary widely. This important area is underresearched, but preliminary findings have confirmed what most practitioners already know: the process of disclosure among children is frequently complex and idiosyncratic.

When planning an interview, police officers and social workers should pay attention to the findings of research that has shown that abused children are very different in the way they disclose. This is important in order to ensure that interviewers tailor their approach to the individual child in question. For example,

- disclosure can be 'accidental' (and non-verbal, e.g. through behaviour) or deliberate (and verbal);
- children, especially young children, are not always the instigators of an allegation or investigation (there may be a medical query, witness reports, or a confession);
- children may not report all details of their abuse at once, and may deliberately withhold or minimise the information they provide;
- children may report their abuse immediately or delay reporting for long periods of time (i.e. years);
- a number of children will deny abuse or later retract their disclosure, even if other evidence exists;
- the presence of a previous informal disclosure does not guarantee a formal
  disclosure in an investigative interview, but there may be a relationship between
  prior disclosure or non-disclosure and disclosure/non-disclosure in the
  investigative interview;
- many factors including age and culture may influence childrens' willingness or ability to disclose abuse; and,
- even if children tell, they may not anticipate professional intervention.

Research has shown that younger children are more likely to make 'accidental disclosures' (i.e. are less likely to have initiated the interview themselves) or to have an unclear 'abuse status' (e.g. Elliott and Briere, 1994). Some children may also consider the consequences of disclosure for themselves, for loved ones, and for the perpetrator, and may restrict the amount of information they give. This indicates that older children in particular should be prepared for the interview and informed about possible outcomes.

## Preparing the child for the interview

If it is not possible to predict exactly how a child will respond in an interview, then certain steps may help to create a context in which a child is more likely to offer an account. The interviewer has an important role, both within the interview itself, and in preparing the child beforehand. To date, the preparation of child witnesses has largely been conceived as preparation for *court* (see section 5). However, research with children and young people who have been involved in child protection investigations has consistently highlighted the need for earlier consideration of the child's position, and specifically, preparation for the investigative interview (and/or medical examination - Berson et al., 1993). Childrens' main concerns include:

- the speed with which an investigation is pursued. Children feel that they have insufficient time to understand what is happening, or to consider whether they are ready to disclose;
- the lack of information about what is happening and the likely outcome of the interview; and,
- the lack of choice about when and where the interview takes place, who accompanies them, and, for some children, the gender of the interviewer.

These issues can be addressed as part of the planning process. As far as possible, children should be prepared for, and involved in decisions about, what will happen in the interview and beyond.

#### Characteristics of the child and family

In planning the interview, a number of factors need to be considered:

# Child age

Practitioners have highlighted the difficulties they face in interviewing very young children within the constraints of the *Memorandum*. These go beyond the types of questions asked and language used. For example, younger children may benefit from using techniques such as drawings, puppets, and dolls. There may, however, be problems with the use of such techniques (see section 4).

Research has not really addressed implications stemming from the age of the child (Poole and Lindsay, 1998). However, studies have highlighted that older children and young people are far from passive participants in interviews (Sas et al., 1995; Wade and Westcott, 1997; Westcott and Davies, 1996b).

#### Child's state of mind

The child's state of mind in the interview may be influenced by a variety of factors. Many of these, including the impact of the actual abuse, will be beyond the control of the interviewer. Although no simple 'syndrome' can account for the effects of sexual or physical abuse (Elliott and Briere, 1994; Kendall-Tackett et al., 1993) and no single symptom is diagnostic of abuse, some symptoms appear more frequently in abused children. These are listed below.

#### Physical abuse

- negative social behaviour, e.g. increased aggression, non-compliance, conduct disorder, criminal activity;
- possible self-injury and suicidal behaviour;
- increased emotional problems, e.g. anxiety, depression, low self-worth; and.
- lower intellectual functioning and academic achievement.

# Sexual abuse

- fears;
- post-traumatic stress disorder;
- behaviour problems;
- sexualized behaviours; and,
- poor self-esteem.

Clearly, diagnosis of some of these symptoms (e.g. post-traumatic stress disorder) is a matter for the relevant professionals, but interviewers need to be sensitive to the symptoms and their implications for the way children may behave at interview. They may also affect perceptions of the child – e.g. in relation to the child's role in the abuse, or their credibility. However, some factors can moderate or exacerbate these effects, such as those relating to the characteristics of the abuse and the child's age. The presence of maternal support throughout the whole process appears to be particularly important for children's well-being where the mother was not involved in, or not party to the abuse (e.g. Elliott and Briere, 1994; Goodman et al., 1992).

Another influence on the child's state of mind at interview is likely to be the degree of stress experienced. Research has concentrated largely on the effects of the

stressful nature of the abuse itself (e.g. Goodman et al., 1991). However, some studies have looked at the influence of a stressful environment on the child's ability to recall or describe an event (e.g. Peters, 1991), or on their willingness to disclose. Yamamoto et al. (1996) found that being suspected of lying was rated by school children internationally as a very stressful event and many abused children have reported that their fear of not being believed discouraged them from disclosing (e.g. Wade and Westcott, 1997). The message for practice, however, is that stress effects are complex (Christianson, 1992). Stress does not automatically negatively affect memory, but being stressed may make it more difficult for children to recall information.

# Disability

Many commentators have lamented the inadequate response of the child protection and criminal justice systems to the abuse of physically and learning disabled children (e.g. Marchant and Page, 1993; Westcott and Cross, 1996). Research on children's abilities to describe experiences includes research with children with mild learning disabilities (e.g. Milne and Bull, 1996) and with deaf children (Porter et al., 1995). These studies have shown that disabled children are able to provide accurate and reliable information in response to open and general questions, that differs little in quality or completeness from that of non-disabled children.

Studies have, however, also reported that disabled children were less accurate in response to (increasingly) specific questions than the non-disabled children. Sigelman et al. (1981) found that 'yes-no' questions were particularly problematic for children who had severe learning difficulties. This poses a dilemma for interviewing those disabled children who need to use communication boards. These are systems of communication based on pictures, symbols or words, depending upon the child's ability; the child communicates by pointing to the appropriate symbols, using their hands or eyes in order to communicate with others (Kennedy, 1992; Marchant and Page, 1997). In addition, some boards may fail to include relevant vocabulary, such as the names of body parts (e.g. 'bottom') and potentially abusive activities (e.g. 'hit'; Marchant and Page, 1993, 1997). Practitioners and some researchers (e.g. Porter et al., 1995) have also suggested that the desire to please the interviewer may be greater in children who are disabled.

## Race and culture

No research has specifically examined eyewitness accounts of children from different racial or cultural groups, although the finding of no difference in accounts has been incidentally reported in studies using participants from different races (e.g.

Gordon et al., 1994). Phillips (1993) and Gupta (1997) argue that the child's experiences (e.g. of different cultural conventions such as the use of sexual terms in conversation with strangers) may influence the professional response to the abuse of that child. Elliott and Briere (1994) found complex patterns in disclosure rates amongst older black, Hispanic and caucasian American children who had been sexually abused. For example, Hispanic children were more likely to have made partial disclosures, whilst black children were more likely to have never disclosed abuse in cases where outside evidence (e.g. medical or confession) was available (e.g. Nagel et al., 1997). A survey of *Memorandum* training (Westcott and Davies, 1996a) found that trainers judged anti-discriminatory practice to be one of weakest components of *Memorandum* training, and that consideration of race and cultural issues was particularly poor.

#### The influence of the interviewer

Studies in the social work domain have reported childrens' and carers' evaluations of interviewers (see reviews by McGee and Westcott, 1996; Wade and Westcott, 1997). More recent research has also explored other aspects of interviewer influence (e.g. parents-as-interviewers; Ricci et al., 1996). One experiment claimed that the most accurate and complete accounts of a staged event were elicited by an interviewer who adopted a more formal or 'business-like' style (Hutcheson et al., 1996). However, factors such as age, type of offence and readiness to disclose might mediate any interviewer effects. Research findings suggest the following implications for the training of interviewers (e.g. Memon et al., 1996; Moston and Engleberg, 1992):

- the child witness desires social support from the interviewer. This includes open body posture, eye contact, smiles, friendly and welcoming demeanour, and has been shown to have a positive effect on children's testimony, particularly in increasing resistance to leading questions. However, care should be taken not to selectively reinforce statements which hint at abuse;
- by contrast, intimidating or non-supportive interviewers can inhibit children's testimony, or increase susceptibility to misleading questions;
- some children may have definite preferences for interviewers of a particular gender, although interviewer gender has also been found to have little overall impact on the quality of the interview; and,
- even when rigorously trained to follow an interviewing protocol, interviewers do not always apply this training consistently when they interview children.

# This section has highlighted that:

- many factors may affect the way in which abuse is disclosed and the time period in which it is disclosed;
- in planning the interview, interviewers should involve children in decisions surrounding the conduct of the interview, e.g. where and when the interview will take place, and whether the interviewer will be male or female;
- a number of factors, including the child's age, state of mind, race and culture, and any impairments they might have, need to be taken into consideration; and,
- interviewers must avoid intimidating the witness and ensure that children are given as much social support as possible.

# 4. Conducting the Memorandum interview

This section considers research on applying the recommendations of the *Memorandum*, as well as innovations in interviewing technique which have occurred since its inception. Some of this research is experimental, but much involves analysis of tapes of *Memorandum* interviews conducted by trained interviewers.

# The phased approach

The Memorandum calls for interviews to progress through a series of phases:

- rapport;
- free narrative:
- open-ended questions;
- closed but specific questions, and,
- closure.

# Rapport

Boggs and Eyberg (1990: 86) define rapport as 'a positive relationship between interviewer and child that sets the tone for the entire assessment process and helps increase both the amount and accuracy of information provided'. Rapport appears to be present in the majority of Memorandum interviews (Davies et al., 1995; Stockdale, 1996). Poorly done, attempts to establish rapport can appear trite and mechanical and some advocate dispensing with rapport altogether when there has been previous contact with the child (Davies et al., 1998). However, the proper use of rapport should permit the interviewer to:

- establish ground rules for the interview and explore the child's understanding of truth and lies;
- help reduce the social distance between interviewer and interviewee;
- estimate the child's level of knowledge and linguistic competence; and,
- establish open-ended questioning as the standard for the interview (Warren et al., 1996).

#### Ground rules

In order to minimise social suggestibility and prevent children from acceding to the views of the interviewer (see section 2), Warren and McGough (1996: 279) suggest challenging the following assumptions of child interviewees:

- 'every question must be answered even though I don't understand';
- 'every question has a right or wrong answer';
- 'the interviewer already knows what happened, so if he or she says something that differs from what I remember, I am wrong'; and,
- 'I am not allowed to answer "I don't know"'.

These assumptions are not regularly challenged by interviewers. Studies suggest that the acceptability of 'don't know' as a response is mentioned by less than half of all interviewers (Davies et al., 1995; Stockdale, 1996). However, experiments where interviewers have emphasised that 'don't know' is an acceptable answer have demonstrated substantial improvements in accuracy and completeness of recall (Mulder and Vrij, 1996; Saywitz et al., 1993b). Interviewers need to clarify these points with child witnesses.

#### Truth and lies

Nursery school-age children have been shown to be capable of identifying examples of true and false statements (Saywitz and Lyon, 1997). However, an understanding of the moral obligation to speak the truth does not normally occur before nine to 10 years (Perner, 1997). Despite this, children are sometimes asked to distinguish between truth and lies, a question which would tax anyone to answer adequately in the abstract (Wood et al., 1996).

As a consequence, the use of appropriate concrete examples is advised. Bekerian and Dennett (1995) point out that saying: 'If I am wearing blue socks and I said I was wearing red socks, would that be the truth or a lie?' involves inaccuracy, but not necessarily an intent to deceive. They therefore recommend examples such as: 'If I knocked over this glass and broke it, and then I told x that you did it, what would I be doing?'.

#### Free narrative

As noted in section 2, free report normally provides the most accurate information from witnesses, but this can be incomplete, especially in younger children. Examination of a sample of *Memorandum* interviews suggested that 28% lacked any free narrative phase (Davies et al., 1995). Securing free narrative is not always easy, but given its value, both in shifting control to the child and as evidence, every effort should be made to obtain it. Procedures which are likely to achieve this include:

- setting appropriate ground rules (such as asking children to give as much detail as they can); and,
- teaching interviewers restraint before interrupting.

## Questioning styles

Research comparing questions in investigative interviews has found that, compared to closed questions, open questions (e.g. 'what did the man look like?/what happened after mummy went shopping?'):

- elicit longer and more detailed answers (e.g. Lamb et al., 1996a);
- produce more accurate information, particularly in younger children (Hutcheson et al., 1996);
- can be repeated without impairing the accuracy of the answers (e.g. Memon and Vartoukian, 1996); and,
- are less likely to elicit ambiguous, non-verbal responses (Hunt et al., 1995).

However, despite the emphasis in the *Memorandum* on the value of open questions, reviews of actual interviews conducted by police officers have found:

- premature use of closed questions (Stockdale, 1996);
- over-use of specific but non-leading questions, with open-ended questions limited to one per interview (Westcott et al., 1998); and,
- only 30% of interviewers beginning with an open-ended question following the free narrative phase (Davies et al., 1995).

It is recognised that many children will not spontaneously disclose information about sensitive issues without the use of direct prompts (Saywitz et al., 1991). This problem is particularly marked when abusers swear children to secrecy or threaten them (Davies and Wilson, 1996). Even when the child is ready to tell their story, specific questions may be necessary to elicit evidential detail, particularly from younger children (Goodman and Schaaf, 1997). In addition, young children may be unskilled in retrieving specific information from memory on the basis of general queries: information about a suspect's appearance may not be readily accessed via a general request (e.g. 'What clothing was he wearing?'), but can be retrieved via a more specific prompt (e.g. 'Do you remember anything about his trousers?') (Hutcheson et al., 1996; Saywitz and Goodman, 1993). However, specific prompts always need to be followed up by a further general question in order to return control of the dialogue to the child (Lamb et al., 1998). An example might be 'where was your sister when all this was happening?' followed by 'what else can you tell me about that visit from your Uncle?'.

# Inappropriate questioning techniques

An analysis of transcripts of American investigative interviews (Hunt et al., 1995) has highlighted some questioning techniques which cause particular difficulties for children:

- forced choice questions (e.g. 'did he have his hands inside or outside your clothes?'); 74% of children respond simply with one or other choice rather than providing their own answer;
- multi-part questions (e.g. 'did he ask you to go upstairs with him and did you

then go in the bedroom?'); when two or more questions were embedded in a single enquiry, 71% of responses did not address all the issues raised; and,

• 'can you...' (e.g. 'can you tell me about that?') questions; questions beginning in this way were more likely to be misunderstood or yield 'yes/no' answers.

As noted previously, the younger the child, the larger the negative impact of these procedures. Another serious problem arises from modifications, as when the interviewer inappropriately rephrases what the child has said (e.g. Roberts and Lamb, 1999). Walker and Hunt (1998) found 74% of a sample of US interview protocols contained such errors and less than 30% were corrected by the children concerned. There is a danger that if children are not given the opportunity to correct interviewer errors, they may inadvertently incorporate them into their own narrative.

#### Closure

The closure stage of the interview should attempt to leave the child in a positive state of mind. According to Cheung (1997), it provides the opportunity for the interviewer to:

- summarise what the child has said;
- answer any questions from the child;
- thank the child for their time and effort;
- provide a contact name and telephone number; and,
- report the end-time of the interview.

Samples of taped interviews suggest that such comprehensive closure is not usually achieved. Closure is often brief to the point of abruptness and key elements are omitted (Davies et al., 1995). In one study of 50 videotaped interviews, 24% of interviewers failed to check important points with the child (Stockdale, 1996).

#### Duration, pace and number of interviews

An initial survey of *Memorandum* interviews suggested that the majority lasted less than one hour (Davies et al., 1995) and subsequent research suggests that most interviews are under this length (Westcott et al., 1998). There is concern among police interviewers that the one hour recommendation is treated as a rule by the courts, and that officers are not always permitted to defend and explain exceptions (Davies et al., 1998). The idea that interviews should never last more than one hour may explain complaints about the pace of some interviews: children in 43% of the interviews observed by Davies et al. (1995) were judged to have been rushed into the questioning phase.

Although the Memorandum allows for multiple interviews, in practice this provision is rarely exercised (Westcott et al., 1998). In some circumstances, multiple interviewing can be a source of suggestive responding, particularly by young children. However experimental research suggests multiple interviews in themselves do not necessarily distort or invalidate an account, provided openended questions are consistently employed (Poole and White, 1993). A second interview may contain material not present in the first or omit previously recalled material, but this inconsistency need not imply unreliability (Memon et al., 1997). This variation is likely to reflect the natural processes associated with retrieval from memory, as there are always more facts available in the memory than we are able to retrieve on any one occasion (Fisher and Cutler, 1995). In the case of child victims of sexual abuse, this additional material may reflect the greater readiness of the child to disclose personal events to an interviewer as rapport develops (Vizard, 1991). It is important to distinguish between multiple interviews which are properly planned and sensitively conducted and repeated interviews, driven by adult priorities and assumptions, rather than the needs of the child. This latter form of repeat interviewing can lead to distorted testimony and have harmful effects upon the child (see section 5, below).

# Assisting child witnesses in the interview

The use of dolls and other props

Children may benefit from access to physical cues to facilitate recall and enactment of events. Younger children in particular might be expected to benefit from such cues given their limited ability to search spontaneously their memories and construct a free narrative account (Pipe et al., 1993). However, it has been argued that the presence of props associated with the alleged offence (e.g. clothing, objects) could either be distracting for the child or encourage a fantasy-based account (Ceci and Bruck, 1993). Considerable research has been conducted on stimulus support in recent years, covering:

<u>Drawings, props and toys:</u> Encouraging a child to draw is often used as a preliminary task to settle the child before the formal interview (Jones, 1992). However, drawing can also be used to assist the child in retrieving information about key events (Burgess and Hartman, 1993). In one experiment, drawing the event helped the verbal recall of five to six year olds but not pre-school children (Butler et al., 1995). Props and toy replicas linked to an incident can also help recall. Experimental studies have demonstrated that:

 physical props are more effective than verbal prompts with children (Wilson and Pipe, 1989);

- this effect is most marked in children below eight years (Pipe et al., 1993);
- the presence of non-relevant props does not increase errors (Ascherman et al., 1997; Pipe and Wilson, 1994); but,
- the use of toys may prompt errors among three to four year old children who do not fully understand the role of symbols (DeLoache, 1990; Salmon et al., 1995).

These encouraging findings from experiments need to be complemented by field studies which explore the use of props and cues in actual investigations (see Hershkowitz et al., 1998).

Anatomically detailed dolls: The use of anatomically detailed dolls in child abuse investigations continues to excite controversy (Ceci and Bruck, 1993; Koocher et al., 1995). There are concerns that they may be suggestive to children leading to false allegations (Bruck et al., 1995) and caution is reflected in current *Memorandum* guidance. Despite their widespread use in clinical interviewing (Levy et al., 1995), there is no agreed design for such dolls (Simkins and Renier, 1996), nor are there accepted protocols as to how they should be used in an investigative interview (Koocher et al., 1995). Part of the difficulty lies in the different ways dolls can be used in interviews. Everson and Boat (1997) list a variety of uses including:

- a conversation starter;
- an anatomical model;
- a demonstration aid;
- a memory aid; and,
- a diagnostic tool.

Detailed research is required on these functions before emphatic recommendations can be made regarding their use in routine interviewing. A recent survey of experiments on the use of dolls by children concluded that older children showed some benefit from the presence of dolls in providing descriptions relating to the touching of private parts, but that they encouraged false reports when combined with suggestive techniques such as leading questions (Everson and Boat, 1997). Doll play in itself cannot be used as a diagnostic tool; while some kinds of play appear more frequent among abused than non-abused children, there is no single interaction with the dolls which is unique to abused children (Koocher et al., 1995). The use of dolls as a communication aid has enjoyed widespread support among clinicians who interview abused children (American Psychological Association, 1991; American Professional Society on the Abuse of Children, 1995). However, from the point of view of the courts, it is important that when the

child appears to demonstrate abuse with the aid of dolls, the interviewer follows this up with verbal questioning to clarify what the child wished to convey (Bull, 1995; Lamb et al., 1996b).

# Alternative questioning techniques

Research has continued on interviewing techniques for child witnesses of abuse.

#### The Cognitive Interview

The Cognitive Interview (CI) was originally developed for use with adult witnesses and has been shown to increase levels of recall relative to traditional interviewing, without a commensurate rise in error (Kohnken et al., 1999). It consists of a package of mnemonic techniques designed to assist witnesses in searching their memory more exhaustively through multiple attempts at recall within a single session (see Memon, 1998, for a recent review). CI has been adapted for use with children, though its use is not advised with those under seven (Memon et al., 1993).

Experimental research has demonstrated positive effects of this form of interview with school-age children (McAuley and Fisher, 1995; Saywitz et al., 1992), and has shown an increased resistance to leading questions (Memon et al., 1996). It has also been adapted successfully for use with children with mild learning difficulties (Milne and Bull, 1996). There are, however, no field studies published on its effectiveness with child witnesses, and the courts have yet to rule on the legal acceptability of such interviews. Prosecutors may have concerns that repeated requests to recall the same events will highlight inconsistencies in the child's account or violate the 'one hour' guideline on interview duration.

#### Semi-scripted interviews

Although research suggests that the most accurate testimony arises from the use of open-ended questions, interviewers seem reluctant to use them. Lamb and Sternberg in America attempted to train interviewers in the use of open-ended prompts, but found that this had little impact on interview style. They therefore proposed the idea of a semi-scripted interview: a series of verbal formulae that interviewers would be required to learn and apply consistently in interviews (Lamb et al., 1998). Thus, interviewers are taught open questioning with an initial invitation such as: 'I gather that something may have happened to you recently which has upset you. Please tell me about that'. They suggest that even a brief answer can be elaborated upon by seizing on a salient feature of what the child has said and offering a further standard prompt: 'Earlier, you mentioned something about a car. Can you tell me everything about that?'. Interviewers are taught to feign confusion as a

strategy (e.g. 'Gosh, I'm confused; you tell me that Grandpa was in your bed. Tell me how that happened').

Field studies in Israel suggest that interviewers trained in the use of semi-scripted interviews increased the amount of free narrative elicited from children by a factor of three compared to their previous methods (Lamb et al., 1997). However, although the semi-scripted interview appears a promising technique, it may need to counter the possible criticisms that it is too inflexible and affords too little initiative to the interviewer. Research is also required to demonstrate that the additional information elicited is accurate and of forensic significance.

#### **SAGE**

SAGE (Systematic Approach to Gathering Evidence) is an approach to interviewing designed for the child who is suspected to have been abused, but who will not talk about it. It involves a series of wide-ranging interviews, each lasting approximately 30 minutes, about significant persons and places in the child's world and their feelings and thoughts about them. Systematic comparisons of the child's responses in these interviews should enable the trained interviewer to identify areas of particular concern (Roberts and Glasgow, 1993). SAGE has proved a useful clinical procedure worthy of further research, but its requirement for repeated interviewing conflicts with existing *Memorandum* recommendations.

This section has reviewed research and practice in conducting *Memorandum* interviews. In doing so, it highlights the need to:

- incorporate as much free-narrative and open-ended questioning as possible into interviews; and,
- avoid inappropriate questioning techniques, such as forced-choice questions, multi-part questions and 'can you' questions.

The section has also discussed the need for the *Memorandum* to provide further guidance on:

- tests of truth and lies;
- the duration, pace and number of interviews;
- the use of drawings, props, toys and anatomically detailed dolls; and,
- specialised interviewing techniques.

# 5. Appearance at court

Although the *Memorandum* is only concerned with the investigative interview, it is important to see the interview as part of a process which culminates, for some children at least, in an appearance at court. Commentators have contrasted (unfavourably) the way children are interviewed, or treated, at court with what is permitted at earlier stages of the investigative process (e.g. Butler, 1997; Kranat and Westcott, 1994). The court appearance itself may be a new, potentially traumatic experience for the child, rather than the satisfactory closure of the investigation (Westcott and Page, 1998). In this section we consider:

- the long-term effects of court appearance; and,
- preparation of child witnesses for court.

## Long-term effects of court appearance

A small number of researchers (e.g. Oates et al., 1995) have attempted to track child victims of sexual abuse through the criminal justice system, and to evaluate the effect of court appearances on them. Methodological difficulties in such research are profound (e.g. sample size and bias, isolating effects of different factors), and research findings have not been straightforward. However, studies suggest that:

- repeated interviews during investigation and pre-trial are detrimental to childrens' well-being;
- testifying more than once has negative effects on children;
- lengthy and harsh cross-examination is harmful;
- support of the child's mother, or a professional such as a guardian *ad litem*, promotes their well-being;
- case resolution (and the passage of time) promotes recovery from the court experience, regardless of outcome; and,
- testifying in court for some children who desire it, can have beneficial effects.

Many researchers have emphasised children's fears about testifying, especially their fear of seeing the defendant, and also the need for adequate preparation of both the child and the lawyers and judges involved in child witness cases. However, overall, it appears that most negative effects eventually dissipate over the long-term. They do, however, persist for a minority of children, depending on factors such as those above, as well as on the child's age and severity of abuse.

#### Preparation of child witnesses for court

Practice perspectives on preparation

The accumulating literature suggests agreement that preparation is a positive and essential provision for child witnesses (e.g. Aldridge and Freshwater, 1993;

Plotnikoff and Woolfson, 1995b; 1996). However, there is also agreement that provision in the UK is currently fragmented, under-resourced and not available to as many as one third of child witnesses (Chandler and Lait, 1996; Davies et al., 1995). There is also the need to challenge assertions (frequently made by defence counsel) that preparation constitutes coaching. The CPS Guidance on Good Practice on Pre-trial Therapy for Child Witnesses (1999) believes, however, that preparation should normally be non-contentious provided there is no attempt to rehearse or discuss the child's evidence.

'The Child Witness Pack' was published by the NSPCC/ChildLine in 1993, and has had a significant impact on practice. A revision – 'The Young Witness Pack' – was launched in 1998 (NSPCC/ChildLine), and includes a new handbook for child witness supporters. Many preparation programmes have been built around The Child Witness Pack, or other similar materials, and are primarily aimed at the children themselves. However, information contained within these packs may also be aimed at carers and adults responsible for preparing children. The main components of most preparation programmes include:

- assessment (e.g. of children's needs and anxieties about testifying);
- education (e.g. teaching children about court procedures, legal terminology);
- preparation on the giving of testimony;
- preparation in order to enhance emotional resilience (stress management);
- involving the child's carer(s);
- liaison and practical arrangements regarding the court process (e.g. the use of video-links); and,
- debriefing (Aldridge, 1997).

If all these components are addressed, both the quality of the child's evidence and the child's well-being are believed to be enhanced. However, Murray (1997) and others have identified some outstanding issues which need to be addressed in relation to preparation programmes:

- who should be responsible for resourcing and delivering preparation?
- should children be prepared in groups or individually?
- what are the training needs of adults who prepare children for court?
- what are the most effective components of preparation programmes?
- are there any unintended side effects of preparation? and,
- how can preparation programmes be tailored to individual needs, e.g. for children who are disabled, as for children who have been subject to different kinds of abuse?

Plotnikoff and Woolfson (1996) evaluated child witness support schemes provided by Victim Support. They drew upon information and interviews from a number of different sources involved with child witnesses. Again, witness support was evaluated positively, though difficulties imposed by other aspects of the criminal justice process (e.g. delays, professional liaison) sometimes worked against these benefits. Children appearing at magistrates courts, and in non-sexual abuse cases, rarely received the provision. The report recommended training of both supporters and members of the judiciary in related issues, and the development of a model preparation protocol.

Experimental perspectives on preparation

Experiments have been carried out with the aim of:

- enhancing children's memory;
- facilitating communication; and,
- helping witnesses resist suggestion.

A variety of techniques have been used. Saywitz et al. (1993) found that:

- Narrative elaboration was a successful technique for increasing the amount of accurate recall offered by children, by prompting or reminding them of the importance of talking about participants, settings, actions, conversations/ emotional states, and consequences. Children aged six to nine years were trained in the use of five pictorial 'cue cards' relating to these features, and produced over 50% more information, without an increase in inaccurate information. The improvements were maintained when questioned by an unfamiliar authority figure.
- Comprehension-monitoring training was successful in helping six to eight year olds identify questions they did not understand, and ask adult interviewers for rephrasing. In the training, children were warned that they may not understand all questions asked of them, and advised as to why adults ask children difficult questions. Training also demonstrated some of the negative consequences of trying to answer questions which were not fully understood. Children receiving this training gave more accurate testimony than comparison groups not receiving the same intervention.
- Resistance training was also successful in enabling seven year old children to identify and respond appropriately i.e. to resist misleading questions put by interviewers. Again, training demonstrated some unanticipated consequences for children who go along with adults' suggestions, as well as teaching children to identify suggestive questions, compare the suggestion to their own memory of the event, and to feel confident in asserting their own response which challenged that of the adult's suggestion.

• Court education and stress inoculation training was successful in increasing the legal knowledge of children aged eight to 10 years, but the increase in knowledge was not associated with a decrease in measured anxiety or improved memory during mock testimony. Training included role play and repeated practice in three sessions, as children were taught about different steps in the investigative process and trial procedures, and related coping skills such as deep breathing and confidence boosting self-statements.

This section has reviewed the effects of appearing in court, highlighting the need for interviewers to:

- minimise the number of repeated interviews during the investigation and pretrial period;
- recognise that lengthy and harsh cross-examination can have negative effects on children; and,
- ensure that children are prepared adequately for their appearance in court.

# 6. Review

What are the implications of the research reviewed in this report for policy and practice? The four areas covered were:

- child development;
- planning the interview;
- conduct of Memorandum interviews; and,
- pre-court preparation.

Here, the broad conclusions of each of these areas are summarised, pointers for practice are discussed and unanswered questions identified. The section concludes with a brief consideration of the wider policy context in which the research needs to be placed.

# Child development

#### Overview

Children as young as three years of age are capable of recalling events they have experienced accurately and with detail, and competence improves with age. What material is remembered will be limited by children's level of conceptual understanding and their ability to search their memory in a systematic and directed way. However, a greater limitation for children of all ages will be the effectiveness of interviewing techniques and the style and form of questions that the interviewer chooses to employ. Questions which are inappropriately structured or which use concepts or vocabulary beyond the grasp of the child will, at best, bemuse, and at worst lead to replies which are misleading and reduce the child's overall credibility. Very young children (those below the age of six) have been identified as particularly vulnerable to suggestive pressures which can on occasion lead to false accounts. Such pressures include:

- denigration of the accused;
- repeated interviews by interviewers committed to a particular version of events; and,
- exposure to misleading accounts.

Research suggests the impact of these factors reduces with age, but given that they can also influence adults' statements, care is required in assessing the testimony of children exposed to such pressures. No reliable method yet exists of judging the truth or otherwise of childrens' allegations, either from their demeanour at interview or from features of their accounts. It is primarily the interviewer's responsibility to ensure, through appropriate training and awareness of childrens' strengths and vulnerabilities at different ages, that the risks of suggestive responding are minimised.

#### Implications for practice

The Memorandum speaks always of 'the child' but the research reviewed on child development suggests that children at different stages of development may require rather different styles of interview. Young children, for instance, will require much more support and encouragement to produce free narrative prior to any specific questions. If the proposals embodied in Speaking Up for Justice (Home Office, 1998) are implemented, it is likely that the numbers of children giving evidence in criminal cases will increase. Any revision of the Memorandum should have a specific section devoted to the interviewing of very young and particularly vulnerable witnesses. Those conducting these interviews require specialist training, above and beyond what is already provided.

The Memorandum includes some general comments on the difficulties of concepts such as time, duration and frequency, but any future edition should incorporate recent findings on the development of these concepts, as well as examples of how questions can be formulated for children of different ages. However, any statement on child age norms needs to be accompanied by a caution that norms cannot be applied rigidly: there will always be children whose intellectual development or emotional state cause them to perform above or below their chronological age. Developmental information about the individual child is crucial to the success of any interview and should be collected at the planning stage, supplemented where necessary by enquiries during the rapport stage of the interview itself.

#### Unanswered questions

Much of the existing experimental research has involved non-abused children. It is important to know what the impact of sexual and physical abuse is upon the general qualities of the abused child's memory. It is known that social and emotional factors influence memory (Eysenck, 1992) and there is evidence that some children who have been victims of abuse may go on to suffer memory failings in adulthood (Lindsay and Read, 1994). Research could assist in identifying subgroups who are particularly resilient or vulnerable to different questioning techniques, as well as tailoring guidance more explicitly toward abused children.

# Planning the interview

#### Overview

The likelihood of interviewers eliciting accurate information from children will be determined not merely by the state of their memory and the phrasing of questions, but also by a range of social, cultural and emotional factors. An investigative interview which does not take these additional factors into account in planning

and preparation is unlikely to be successful. Children may have had threats made against them, may have entered secrecy pacts or have conflicting feelings toward the abuser, all of which can lead them to minimise or limit their accounts of abuse. In some instances, evidence of sexual or physical abuse may have arisen from third parties – siblings, teachers or medical practitioners – and children may need both time and reassurance before being able to disclose what has happened to them in a formal interview. Finally, interviewers need also to be aware of the disruption to emotions and behaviour that can characterise victims of child abuse and be prepared to accommodate this. Above all, interviewers need to generate sufficient warmth and mutual trust to permit frank disclosure of sensitive issues, while at the same time ensuring that questioning is conducted in an evidentially sound manner. Such an interview style requires experience and rigorous training to develop and maintain.

#### Implications for practice

Interviewers need to collate as much information as possible regarding the child and their family background. This includes considering issues such as the child's age, their state of mind at the time of the interview, their race and culture, and any impairments they might have. Research on the ways in which children may disclose also needs to be considered. A successful interview is also more likely to arise if children are enabled to take informed decisions covering such matters as the gender of the interviewer, and the time and location of the interview. Children also need to know what the consequences of the interview may be, both for themselves and the abuser – frankness from the child requires frankness from the interviewer. Legally acceptable pre-interview protocols need to be developed, covering the purpose of the interview and establishing any particular requirements the child may have.

# Unanswered questions

More research is needed on the process by which a child's allegation turns into a *Memorandum* interview and finally into an appearance at court. Outcomes for children who undergo a *Memorandum* interview – and perhaps nothing further – are also in need of investigation. Evidence suggests that the decision-making at each stage can be inconsistent. Explicit criteria advising when an interview should be undertaken should be developed by the Home Office in consultation with interested parties, and included in the *Memorandum*. A further area of concern is the timing of any interview. Evidential considerations may demand an early interview, but this will be counter-productive if the child is not ready to disclose: much more needs to be learned about the optimal timing of interviews, which could assist in the development of pre-interview protocols.

#### Conducting the Memorandum interview

#### Overview

Interviews conducted under the Memorandum generally elicit an account of the child's story and employ styles of questioning which are acceptable to the courts. Research confirms that open-ended questions yield more information, and of greater accuracy, than closed questions. However, those conducting Memorandum interviews demonstrate insufficient emphasis on the use of open-ended questioning and free narrative from the child. Open-ended questions do not always precede closed questions and can be too limited in number. The ideal endorsed by the Memorandum of a simple progression from open through ever more specific questions is rarely achieved in practice. This is not surprising, given the sensitive nature of the material explored in investigative interviews and the limited narrative powers of younger or anxious children. Evidence gathering often requires recurrent questioning as one incident is explored and then another. This should not be seen as compromising the integrity of an interview as long as questioning on each incident begins with open-ended questions and the initiative remains with the child. Much can be done to encourage free narrative by children through proper use of the rapport phase.

#### Implications for practice

Given the range of children involved, in terms of age, culture, disability and emotional state, there is a strong case for emphasising the principles underlying sound evidential interviewing, rather than being over-prescriptive about one particular model. This applies in particular to advice on the progression of interview phases and the length of an interview. Given that an ideal progression cannot and will not be achieved in most interviews, any revision of the *Memorandum* should stress:

- the evidential value of responses to open-ended questions;
- the importance of following closed questions with further open-questions; and,
- the importance of giving the initiative for responding to the child.

This should be supplemented by examples of productive and counter-productive questioning techniques of the kind illustrated in section 4. One of the criticisms levelled against the existing *Memorandum* is the lack of sufficient examples, and there is now a fund of useful formulations available. Given its use as a core document for training courses (Davies et al., 1998), an extended range of concrete examples should always supplement general advice. In addition, if the recommended maximum interview length of one hour is to be retained, then the

guidance needs to acknowledge that there will be exceptions. Obvious examples include children with communication difficulties or cases where multiple incidents of abuse are being explored.

The importance of developmental changes in childrens' memory and susceptibility to suggestible questioning have already been emphasised. However, childrens' performance at interview not only reflects the questions they are asked, but also their own perceptions of the interview process. The existing *Memorandum* provides some recognition of these concerns through the inclusion of the rapport and closure stages, but it is evident from research that the function of these phases is insufficiently appreciated in practice. Any revision of the *Memorandum* needs to stress the importance of establishing and addressing children's needs and concerns prior to interview and highlighting the role of appropriate rapport and closure in that process.

The *Memorandum* also recognises that disabled children and those from minority ethnic communities will require special consideration, but goes little further than acknowledging the problem. As the literature makes clear, children with communication or learning impairments have difficulty in accommodating the requirements of the *Memorandum* interview with its emphasis upon free report. These difficulties and the need for flexibility in styles of interviewing such children need to be spelled out explicitly in any new edition, and clear guidance given as to where information and expertise may be found.

In one area, the existing *Memorandum* may be insufficiently prescriptive. When the legislation was introduced, it was envisaged that all children who were witnesses to sexual and physical violence, whether as victims or bystanders, would be eligible for *Memorandum* interviews. As surveys have repeatedly demonstrated, *Memorandum* interviews are conducted overwhelming in sexual abuse cases, much less frequently in physical abuse cases, and rarely in instances where children are witnesses to domestic violence or street crime (CPS Inspectorate, 1998; Wade, 1997). Given the specialised nature of the interviewing procedures required to elicit accurate testimony from children, it should be routine practice for all such interviews to be conducted by *Memorandum* trained officers, and any revision should emphasise its wider application.

# Unanswered questions

There appear to be different interviewing styles, even within the confines of the *Memorandum* interview. How the styles differ from each other and what their impact is upon children are unknown and merit further evaluation. Clearly, if

certain styles are particularly facilitative, then these could be emphasised in training. It would also be advantageous to offer specific recommendations on the provision of social support in interviews and on the use of stimulus support (drawings, props, toys and models). Guidance on alternative interviewing techniques would also be useful.

With changes in attitude and legislation, certain sub-groups of children are likely to figure more prominently as interviewees. These include:

- young children (e.g. six years and under);
- disabled children; and,
- children from ethnic communities.

Research into these particular groups of children may be necessary in the future, along with the development of special questioning techniques.

#### Appearance at court

#### Overview

Research demonstrates that giving evidence at court is a traumatic experience for most children, though the long-term consequences appear less clear-cut. Research also shows that the effects on the child of court appearance can be ameliorated through careful pre-court preparation. The value of preparation was recognised through official support for *The Child Witness Pack* (1993) and *The Young Witness Pack* (1998). Existing research has identified some categories of child witness who are especially vulnerable – e.g. those whose first language is not English, disabled children, young children, and those who lack parental support – and it is important that these children receive particular attention. In addition to better preparation, much can be done to make giving of evidence a less threatening experience. The provision of live links and the use of videotaped interviews are examples, and the proposals incorporated in the 1999 *Youth Justice and Criminal Evidence Bill* would carry this a stage further.

#### Implications for practice

Pre-court preparation appears to have benefits for the children involved. However, provision is uneven across court centres and there is a need for national coordination and uniform standards. This needs to extend to all levels of court, given that 20% of children involved in abuse cases give their evidence in magistrates' courts which are not normally equipped with video or other special facilities (Home Office, 1998). Some comments on the value and legal acceptability of precourt preparation should be included in a revised *Memorandum*: the effort of careful interviewing can be wasted if the child is unprepared for examination at court.

#### Unanswered questions

Preparation covers a multitude of procedures and it is not clear which components are likely to be most helpful; more research is required to examine the value of individual elements of the preparation package. Some techniques, particularly those relating to the searching of memory under stress and teaching resistance to leading or poorly understood questions have produced promising results in the laboratory. Provided that such procedures prove legally acceptable, their impact needs to be explored under courtroom conditions. Other aspects of court appearance require further investigation – such as case outcomes and childrens' experiences subsequent to the trial, and the role of barristers and judges in influencing childrens' experiences of testifying.

# Concluding comments

Sentiment on the competency of child witnesses has been subject to pendulum swings in recent years. Headlines proclaim that children are either very good witnesses or very dangerous witnesses; children never lie about abuse or they lie easily about it; the law is tipped overwhelmingly in favour of the defendant or it accedes unnecessarily to the whims of the child. The research covered in this review has attempted to go beyond simplistic arguments to look at the implications of research findings. In many areas, however, the necessary research does not exist and there is a need for more systematic studies to address the tangle of social, developmental and motivational factors which determine the accuracy and suggestibility of a given child under questioning. However, enough findings of substance have emerged to underpin a revision of the *Memorandum*.

The main implication of the research outlined here is a move away from a single prescribed questioning style toward a more flexible approach based upon broad principles of interviewing, taking account of childrens' ages, state and abilities, their social and cultural background and the nature of the allegations. Such an approach requires more planning and individual decision-making by the interviewer, but would enable a wider range of children to qualify for a formal interview. It would also protect interviewers from being legally impaled on what many defence lawyers continue to perceive as 'the rules' of the existing Memorandum, while still allowing the courts to reject improper interviews. This approach also has implications for training, which again would require greater insight on the part of the trainee and less adherence to formulae. The introduction of a nationally accepted training curriculum is recommended (Davies et al., 1998).

Revision of the *Memorandum* will take place within a context determined by developments in broader policy on investigating child abuse. The Home Office's

proposals in *Speaking up for Justice* (1998) echo Sir William Utting (Utting et al., 1997) in calling for videotaped cross-examination as well as examination as originally proposed by Pigot (1989), avoiding the need for children to appear in court at all. The 1999 *Youth Justice and Criminal Evidence Act* will incorporate such procedures into law. However, experience in Scotland and Western Australia, where similar legislation already exists, suggests there is a reluctance on the part of court professionals to take full advantage of such facilities, and children continue to be cross-examined in court in the conventional way (see Davis et al., 1999, for an overview of provisions in other jurisdictions). The Department of Health's consultative document on *Working Together* (1998) foresees a more pro-active role for social workers, but contains no suggestion that they withdraw from involvement in *Memorandum* interviewing. There will, however, be increasing concern among some child protection professionals that the social and emotional needs of the child witness highlighted in this report might not be addressed if the drift towards exclusively police-led interviewing continues (Brownlow and Waller, 1997).

No current policy initiative is, however, likely to curtail investigative interviewing, or to end its link to the criminal justice system. This link forms the weakest part of the chain joining the child complainant to the court. For example, as has been noted, there are no explicit national criteria determining when an interview is made, or thresholds which must be exceeded before an interview becomes evidence likely to lead to prosecution. In order to have any impact on policy, a revision of the *Memorandum* needs to be geared to the introduction of nationally agreed decision rules to replace the current variable system.

# References\*

**Aldridge, J.** (1997) Next steps after the Memorandum: Preparing children for court. In H.L. Westcott and J. Jones (Eds), Perspectives on the Memorandum. Policy, Practice and Research in Investigative Interviewing. Aldershot, Hants: Arena.

Aldridge, J. and Freshwater, K. (1993) The preparation of child witnesses. *Tolley's Journal of Child Law*, 5, 25-28.

American Professional Society of the Abuse of Children. (1995) Practice Guidelines: Use of Anatomical Dolls in Child Sexual Abuse Assessment. Chicago: APSAC.

American Psychological Association. (1991) Statement on the use of anatomically detailed dolls in forensic evaluations. APA Council of Representatives. Unpublished manuscript.

**Aschermann, E., Dannenberg, U., and Schultz, A-P** . (1997) 'Photographs as retrieval cues for children.' *Applied Cognitive Psychology*, 11.

Baker-Ward, L., Gordon, B.N., Ornstein, P. A., Larus, D.M. and Clubb, P.A. (1993) 'Young children's long-term retention of a pediatric examination.' *Child Development*, 64, 1519-1533.

**Barker, B., Jones, J., Saradjian, J. and Wardell, R** . (1998) Abuse in Early Years: Report of the Independent Inquiry into Shieldfield Nursery and Related Events. Newcastle: Newcastle City Council.

**Bekerian, D.A.** and **Dennett, J**. (1992) 'The truth in content analyses of a child's testimony.' In F. Losel, D. Bender and T. Bliesner (Eds), *Psychology and Law – International Perspectives*. Berlin: De Gruyter.

**Bekerian, D.A. and Dennett, J.** (1995) The Child's Account. Evidential Interviews with Children: Observations and Recommendations on Video Procedures and Interview Techniques. Cambridge: Cambridgeshire County Council.

**Bernard. C.** (1995) A study of black mothers' emotional and behavioural responses to the sexual abuse of their children. Paper presented to the 4th International Family Violence Research Conference, University of New Hampshire, Durham, July 21-24.

Berson, N.L., Herman-Giddens, M.E. and Frothingham, T.E. (1993) 'Children's perceptions of genital examinations during sexual abuse evaluations.' *Child Welfare*, 72, 41-49.

<sup>\*</sup> This list also includes references not cited in the text. These are intended to provide sources of reference to other useful texts and papers.

Boggs, S.R. and Eyberg, S.E. (1990) 'Interview techniques and establishing rapport.' In A.M. La Greca (Ed), *Through the Eyes of the Child: Obtaining Self-Reports from Children and Adolescents*. Boston: Allyn and Bacon.

**Bradford, R.** (1994) 'Developing an objective approach to assessing allegations of sexual abuse.' *Child Abuse Review*, 3, 93-101.

**Bradley, A. R. and Wood, J.M.** (1996) 'How do children tell? The disclosure process in child sexual abuse'. *Child Abuse and Neglect*, 20, 881-891.

Brainerd, C.J., Reyna, V.F., Howe, M.L. and Kingma, J. (1990) 'The development of forgetting and reminiscence.' Monographs of the Society for Research in Child Development, 55 (3-4) (Serial No. 222).

**Brownlow, J. and Waller, B**. (1997) 'The Memorandum: A social services perspective.' In H.L. Westcott and J. Jones (Eds), *Perspectives on the Memorandum. Policy, Practice and Research in Investigative Interviewing*. Aldershot, Hants: Arena.

Bruck, M., Ceci, S.J., Francoeur, E. and Renick, A. (1995) 'Anatomically detailed dolls do not facilitate preschoolers' reports of a pediatric examination involving genital touching.' *Journal of Experimental Psychology: Applied*, 1, 95-109.

**Bull, R.** (1995) 'Good practice for video recorded interviews with child witnesses for use in criminal proceedings.' In G. Davies, S. Lloyd-Bostock, M. McMurran and C. Wilson (Eds), *Psychology, Law and Criminal Justice: International Developments in Research and Practice.* Berlin: De Gruyter.

**Burgess, A. W. and Hartman, C.R**. (1993) 'Children's drawings.' Child Abuse and Neglect, 17, 161-172.

Butler, A. (1993) 'Spare the child.' Police Review, 141-151.

**Butler, A.** (1997) 'The Memorandum: The police view.' In H.L. Westcott and J. Jones (Eds), *Perspectives on the Memorandum. Policy, Practice and Research in Investigative Interviewing.* Aldershot, Hants: Arena.

Butler, S., Gross, J. and Hayne, H. (1995) 'The effect of drawing on memory performance in young children.' *Developmental Psychology*, 31, 597-608.

Campis, L.B., Hebden-Curtis, J. and Demaso, D.R . (1993) 'Developmental differences in detection and disclosure of sexual abuse.' *Journal of the American Academy of Child and Adolescent Psychiatry*, 32, 920-924.

**Carter, C.A., Bottoms, B.L. and Levine, M.** (1996) 'Linguistic and socioemotional influences on the accuracy of children's reports.' *Law and Human Behavior*, 20, 335-358.

**Cassel, W. and Bjorklund, D.** (1995) 'Developmental patterns of eyewitness responses to repeated and increasingly suggestive questions.' *Law and Human Behavior*, 19, 507-531.

Ceci, S.J. and Bruck, M. (1993) 'Suggestibility of the child witness: A historical review and synthesis.' *Psychological Bulletin*, 113, 403-439.

**Ceci, S.J. and Bruck, M.** (1995) *Jeopardy in the Courtroom:* A Scientific Analysis of Children's Testimony. Washington, D.C.: American Psychological Association.

**Ceci, S.J., Crotteau Huffman, M.L. and Smith, E.** (1994) 'Repeatedly thinking about a non-event: Source misattributions among preschoolers.' *Consciousness and Cognition*, 3, 388-407.

**Chandler, J. and Lait, D.** (1996) 'An analysis of children as witnesses in the crown court.' In Victim Support (Ed), *Children in Court*. London: Victim Support.

**Cheung, K-F.** (1997) 'Developing the interview protocol for video-recorded child sexual abuse investigations: A training experience with police officers, social workers and clinical psychologists in Hong Kong'. *Child Abuse and Neglect*, 21, 273-284.

**Christianson, S-A.** (1992) 'Emotional stress and eyewitness memory: A critical review.' *Psychological Bulletin*, 112, 284-309.

**Clarke, C.** (1994) The Effect of Police Officers' Language When Interviewing Children. London: Home Office Police Research Award Scheme Series.

**Crown Prosecution Service Inspectorate.** (1998) The Inspectorate's Report on Cases Involving Child Witnesses. London: Crown Prosecution Service Inspectorate.

**Crown Prosecution Service.** (1999) Pre-trial Therapy for Child Witnesses – Good practice guidance. Draft (unpublished).

**Davies, G.M.** (1992) 'Influencing public policy on eyewitnessing: Problems and possibilities.' In F. Losel, D. Bender and T. Bliesner (Eds), *Psychology and Law: International Perspectives*. Berlin: De Gruyter.

**Davies, G.M., Marshall, E. and Robertson, N**. (1998) *Child Abuse: Training Investigating Officers*, Police Research Series Paper 94, London: Home Office Policing and Reducing Crime Unit.

**Davies, G.M. and Noon, E.** (1991) An Evaluation of the Live Link for Child Witnesses. London: Home Office.

**Davies, G.M. and Westcott, H.L.** (1995) 'The child witness in the courtroom: Empowerment or protection?' In M.S. Zaragoza, J.R. Graham, G.C.N. Hall, R. Hirschman and Y.S. Ben-Porath (Eds), *Memory and Testimony in the Child Witness*. Thousand Oaks, CA: Sage Publications.

**Davies, G.M. and Wilson, J.C.** (1996) 'Child witnesses – Three years on.' Forensic Update, 46, 20-22.

Davies, G.M., Wilson, C., Mitchell, R. and Milsom, J. (1995) Videotaping Children's Evidence: An Evaluation. London: Home Office.

Davis, G., Hoyano, L., Keenan, C., Maitland, L. and Morgan, R . (1999) An assessment of the admissibility and sufficiency of evidence in child abuse prosecutions, Home Office Occasional Paper, London: Home Office (in press).

**DeLoache, J.S.** (1990) 'Young children's understanding of models. In R. Fivush and J. Hudson (Eds),' *Knowing and Remembering in Young Children*. New York: Cambridge University Press.

**DeLoache, J.S. and Marzolf, D.P.** (1995) 'The use of dolls to interview young children: Issues of symbolic representation.' *Journal of Experimental Child Psychology*, 60, 155-173.

**Dent, H.R.** (1986) 'An experimental study of the effectiveness of different techniques of questioning mentally handicapped child witnesses.' *British Journal of Clinical Psychology*, 25, 13-17.

**Department of Health.** (1998) Working Together to Safeguard Children: New Government Proposals for Inter-Agency Co-operation. Consultation Paper. London: Department of Health.

**Dezwirek -Sas, L., Hurley, P., Austin, G. and Wolfe, D.** (1991) Reducing the System-Induced Trauma for Child Sexual Abuse Victims through Court Preparation, Assessment and Follow-Up. Ontario: London Family Court Clinic.

**Doris, J.** (1991) The Suggestibility of Children's Recollections: Implications for Eyewitness Testimony. Washington, D.C.: American Psychological Association.

**Elliott, D.M. and Briere, J.** (1994) 'Forensic sexual abuse evaluations of older children: disclosures and symptomatology.' *Behavioral Sciences and the Law*, 12, 261-277.

**Everson, M.D.** (1997) 'Understanding bizarre, improbable, and fantastic elements in children's accounts of abuse.' *Child Maltreatment*, 2, 134-149.

**Everson, M.D. and Boat, B.W**. (1997) 'Anatomical dolls in child sexual abuse assessments: A call for forensically relevant research.' *Applied Cognitive Psychology*, 11, 55-74.

Eysenck, M.W. (1992) Anxiety: The Cognitive Perspective. Hove: Erlbaum.

**Fisher, R.P. and Cutler, B.L.** (1995) 'The relation between consistency and accuracy of eyewitness testimony.' In G. Davies, S. Lloyd-Bostock, M. McMurran and C. Wilson (Eds), *Psychology, Law and Criminal Justice: International Developments in Research and Practice.* Berlin: De Gruyter.

Flin, R.H., Boon, J., Knox, A. and Bull, R. (1992) 'The effect of a five-month delay on children's and adults' eyewitness memory.' *British Journal of Psychology*, 83, 323-335.

Flin, R.H., Stevenson, Y. and Davies, G.M. (1989) 'Children's knowledge of court proceedings.' *British Journal of Psychology*, 80, 285-297.

**Fundudis, T.** (1997) 'Young children's memory: How good is it? How much do we know about it?' Child Psychology and Psychiatry Review, 2, 150-158.

Goodman, G.S., Bottoms, B.L., Schwartz-Kenney, B.M. and Rudy, L . (1991) 'Children's testimony for a stressful event: Improving children's reports.' *Journal of Narrative and Life History*, 1, 69-99.

Goodman, G.S. and Schaaf, J.M. (1997) 'Over a decade of research on children's eyewitness testimony: What have we learned? Where do we go from here?' *Applied Cognitive Psychology*, 11, 5-20.

Goodman, G.S., Sharma, A., Thomas, S.F. and Considine, M.G. (1995) 'Mother knows best: Effects of relationship status and interviewer bias on children's memory.' *Journal of Experimental Child Psychology*, 60, 195-228.

Goodman, G.S., Taub, E.P., Jones, D.P.H., England, P., Port, L.K., Rudy, L. and Prado, L. (1992) 'Testifying in criminal court: Emotional effects on child sexual assault victims.' *Monographs of the Society for Research in Child Development*, 57(5), serial no. 229.

Gordon, B.N., Jens, K.G., Hollings, R. and Watson, T.E. (1994) 'Remembering activities performed versus those imagined: Implications for testimony of children with mental retardation.' *Journal of Clinical Child Psychology*, 23, 239-248.

**Gupta, A.** (1997) 'Black children and the Memorandum.' In H.L. Westcott and J. Jones (Eds), *Perspectives on the Memorandum. Policy, Practice and Research in Investigative Interviewing*. Aldershot, Hants: Arena.

Hershkowitz, I., Orbach, Y., Lamb, M.E., Sternberg, K.J. Horowitz, D. and Hovav, M. (1998) 'Visiting the Scene of the Crime: Effects on children's recall of alleged abuse', Legal and Criminological Psychology, 3, 195-208.

**Holton, J. and Bonnerjea, L.** (1994) The Child, the Court and the Video: A Study of the Implementation of the Memorandum of Good Practice on Video Interviewing of Child Witnesses. London: Department of Health Social Services Inspectorate.

**Home Office** . (1998) Speaking Up for Justice: Report of the Interdepartmental Working Group on the Treatment of Vulnerable or Intimidated Witnesses in the Criminal Justice System. London: Home Office Procedures and Victims Unit.

Home Office in conjunction with Department of Health. (1992) Memorandum of Good Practice on Video Recorded Interviews with Child Witnesses for Criminal Proceedings. London: HMSO.

Horowitz, S.W., Lamb, M.E., Esplin, P. W., Boychuk, T.D., Krispin, O. and Reiter-Lavery, L. (1999) 'Reliability of criteria-based content analysis of child witness statements.' *Legal and Criminological Psychology* (in press).

**Huffman, M.L.** (1998) Individual differences in children's suggestibility for morally charged events. Poster presented at American Psychology-Law Society Biennial Conference, Redondo Beach, CA, March 6.

- **Huffman, M.L., Warren, A. and Frazier, S**. (1998) 'Truth/lie discussions in sexual abuse interviews with children: 'The truth or not the truth?' Paper presented to the American Psychology-Law Society Biennial Conference, Redondo Beach, CA, March 6.
- **Hughes, B., Parker, H. and Gallagher, B** . (1996) *Policing Child Sexual Abuse: The View from Police Practitioners.* London: Home Office Police Research Group.
- Hunt, J., Komori, L., Kellen, L., Gallas, J. and Gleason, T . (1995) Faulty and non-productive questioning techniques: Potential pitfalls of the child interview. Paper presented to the Meeting of the Society for Applied Research in Memory and Cognition, Vancouver, BC, July 16.
- **Hutcheson, G., Baxter, J.S., Telfer, K. and Warden, D** . (1996) 'Child witness statement quality: Question type and errors of omission.' *Law and Human Behavior*, 19, 631-648.
- **Hyman, I.E., Husband, T.H. and Billings, F.J** . (1995) 'False memories of childhood experiences.' *Applied Cognitive Psychology*, 9, 181-197.
- **Jones, D.P.H.** (1992) Interviewing the Sexually Abused Child: Investigation of Suspected Abuse. London: Gaskell.
- **Keary, K. and Fitzpatrick, C.** (1994) 'Children's disclosure of sexual abuse during formal investigation.' *Child Abuse and Neglect*, 18, 543-548.
- **Kellogg, N.D. and Huston, R.L.** (1995) 'Unwanted sexual experiences in adolescents: Patterns of disclosure.' *Clinical Pediatrics*, 34, 306-312.
- **Kennedy, M. (1992)** 'Not the Only Way to Communicate: A challenge to voice in child protection work', *Child Abuse Review*, 1, 169-177.
- Kendall-Tackett, K.A., Williams, L.M. and Finkelhor, D . (1993) 'Impact of sexual abuse on children: A review and synthesis of recent empirical studies.' *Psychological Bulletin*, 113, 164-180.
- **Koehnken, G., Milne, R., Memon, A. and Bull, R** . (1999) 'The cognitive interview: A meta-analysis.' *Psychology, Crime and Law,* 5, 3-27.

Koocher, G., Goodman, G.S., White, C., Friedrich, W., Sivan, A. and Reynolds, C. (1995) 'Psychological science and the use of anatomically detailed dolls in child sexual abuse assessments.' *Psychological Bulletin*, 118, 199-222.

**Kranat, V.K. and Westcott, H.L.** (1994) 'Under fire: Lawyers questioning children in criminal courts.' *Expert Evidence*, 3, 16-24.

Lamb, M.E., Hershkowitz, I., Sternberg, K.J., Esplin, P.W., Hovav, M., Manor, T. and Yudilevitch, L. (1996a) 'Effects of investigative style on Israeli children's responses.' *International Journal of Behavioral Development*, 19, 627-637.

Lamb, M.E., Hershkowitz, I., Sternberg, K.J., Boat, B. and Everson, M.D. (1996b) Investigative interviews of alleged sexual abuse victims with and without anatomical dolls. *Child Abuse and Neglect*, 20, 1239-1247.

Lamb, M.E., Sternberg, K.J. and Esplin, P.W . (1998) 'Conducting investigative interviews of alleged sexual abuse victims.' Child Abuse and Neglect, 22, 813-823.

Lamb, M.E., Sternberg, K.J., Esplin, P. W., Hershkowitz, I., and Orbach, Y. (1997) 'Assessing the credibility of children's allegations of sexual abuse: A survey of recent research.' Learning and Individual Differences, 9, 175-194.

Lamb, M.E., Sternberg, K.J., Esplin, P.W., Hershkowitz, I., Orbach, Y. and Hovav, M. (1997) 'Criterion-based content analysis: A field validation study.' *Child Abuse and Neglect*, 21, 255-264.

**Leichtman, M. and Ceci, S. J.** (1995) 'The effects of stereotypes and suggestions on preschoolers' reports.' *Developmental Psychology*, 31, 568-578.

**Lepore, S. and Sesco, B.** (1994) 'Distorting children's reports and interpretations of events through suggestion.' *Journal of Applied Psychology*, 79, 108-120.

Levy, H., Markovic, J., Kalinowski, M., Ahart, S. and Torres, H. (1995) 'The use of anatomical dolls and the reliability of information.' *Journal of Interpersonal Violence*, 10, 334-353.

**Lindsay, D.S. and Read, J. D.** (1994) 'Psychotherapy and memories of child sexual abuse: A cognitive perspective.' *Applied Cognitive Psychology*, 8, 281-338.

**Marchant, R. and Page, M.** (1993) Bridging the Gap. Child Protection Work with Children with Multiple Disabilities. London: NSPCC.

Marchant, R. and Page, M. (1997) 'The Memorandum and disabled children.' In H. Westcott and J. Jones (Eds), *Perspectives on the Memorandum. Policy, Practice and Research in Investigative Interviewing*. Aldershot, Hants: Arena.

McAuley, M. and Fisher, R.P. (1995) 'Facilitating children's eyewitness recall with the revised cognitive interview.' *Journal of Applied Psychology*, 80, 510-516.

McAuliff, B., Kovera, M.B. and Viswesvaran, C. (1998) Methodological issues in child suggestibility research: A meta-analysis. Paper presented to the American Psychology-Law Society Biennial Conference, Redondo Beach, CA, March 6.

McGee, C. and Westcott, H.L. (1996) 'System Abuse: Towards a greater understanding from the perspectives of children and parents', *Child and Family Social Work*, 26, 451-474.

**Memon, A.** (1998) 'Telling it all: The cognitive interview.' In A. Memon, A. Vrij and R. Bull (Eds), *Psychology and Law: Truthfulness*, *Accuracy and Credibility*. Maidenhead: McGraw-Hill.

Memon, A., Cronin, O., Eaves, R. and Bull, R . (1993) 'The cognitive interview and child witnesses.' In N.K. Clark and G.M. Stephenson (Eds), *Children, Evidence and Procedure*. Leicester: British Psychological Society.

**Memon, A. and Vartoukian, A.** (1996) 'The effects of repeated questioning on young children's eyewitness testimony.' *British Journal of Psychology*, 87, 403-415.

**Memon, A., Vrij, A. and Bull, R**. (1998) Psychology and Law: Truthfulness, Accuracy and Credibility. Maidenhead: McGraw-Hill.

Memon, A., Wark, L., Holley, A., Bull, R. and Koehnken, G. (1996) 'Interviewer behaviour in investigative interviews.' *Psychology*, *Crime and Law*, 3, 135-155.

Merritt, K.A., Ornstein, P.A. and Spicker, B . (1994) 'Children's memory for a salient medical procedure: Implications for testimony'. *Pediatrics*, 94, 17-23.

Milne, R. and Bull, R. (1996) 'Interviewing children with mild learning disability with the cognitive interview.' In N.K. Clark and G.M. Stephenson (Eds), *Investigative and Forensic Decision Making*. Leicester: British Psychological Society.

**Moston, S. and Engleberg, T.** (1992) 'The effects of social support on children's eyewitness testimony.' *Applied Cognitive Psychology*, 6, 61-75.

**Mulder, M.R.** and **Vrij, A.** (1996) 'Explaining conversation rules to children: An intervention study to facilitate children's accurate responses.' *Child Abuse and Neglect*, 21, 137-147.

Murray, K. (1997) Preparing Child Witnesses for Court. Edinburgh: The Stationery Office.

Nagel, D.E., Putnam, F.W., Noll, J.G. and Trickett, P.K. (1997) 'Disclosure patterns of sexual abuse and psychological functioning at a 1-year follow-up.' Child Abuse and Neglect, 21, 137-147.

**Newcombe, P. and Siegal, M.** (1997) 'Explicitly questioning the nature of suggestibility in preschoolers' memory and retention.' *Journal of Experimental Child Psychology*, 67, 185-203.

NSPCC/ChildLine. (1993) The Child Witness Pack. London: NSPCC.

NSPCC/ChildLine. (1998) The Young Witness Pack. London: NSPCC.

Oates, R.K., Lynch, D.L., Stern, A.E., O'Toole, B.I. and Cooney, G . (1995) 'The criminal justice system and the sexually abused child. Help or hindrance?' *The Medical Journal of Australia*, 162, 126-130.

Office for National Statistics (1991) 1991 Census Report for Great Britain, London: HMSO.

Ornstein, P.A., Baker -Ward, L., Gordon, B.N. and Merritt, K.A. (1997) 'Children's memory for medical experiences: Implications for testimony.' *Applied Cognitive Psychology*, 11, S87-S104.

Ornstein, P.A., Gordon, B.N. and Larus, D. (1992) 'Children's memory for a personally experienced event: Implications for testimony'. *Applied Cognitive Psychology*, 6, 49-60.

**Perner, J.** (1997) 'Children's competency in understanding the role of a witness: Truth, lies and moral ties.' *Applied Cognitive Psychology*, 11, 21-35.

**Peters, D.** (1991) 'The influence of stress and arousal on the child witness.' In J.Doris (Ed), *The Suggestibility of Children's Recollections: Implications for Eyewitness Testimony*. Washington, D.C.: American Psychological Association.

Petronio, S., Reeder, H.M., Hecht, M.L. and Ros-Mendoza, T. (1996) 'Disclosure of sexual abuse by children and adolescents.' *Journal of Applied Communication Research*, 24, 181-199.

**Pezdek, K. and Roe, C.** (1997) 'The suggestibility of children's memory for being touched: Planting, erasing and changing memories.' *Law and Human Behavior*, 21, 95-106.

Pezdek, K., Finger, K. and Hodge, D. (1998) 'Planting false memories: The role of event plausibility.' *Psychological Science*, 8, 437-441.

**Phillips, M.** (1993) 'Investigative interviewing: Issues of race and culture.' In W. Stainton-Rogers and M. Worrel, (Eds), *Investigative Interviewing with Children*. Milton Keynes: The Open University.

**Pigot, T.** (1989) The Report of the Advisory Group on Video Evidence. London: Home Office.

**Pipe, M-E., Gee, S. and Wilson, C.** (1993) 'Cues, props and context: Do they facilitate children's event reports?' In G. Goodman and B. Bottoms (Eds), *Child Victims, Child Witnesses: Understanding and Improving Testimony*. New York: Guilford Press.

**Pipe, M-E. and Wilson, C.** (1994) 'Cues and secrets: Influences on children's event reports.' *Developmental Psychology*, 30, 515-525.

**Plotnikoff, J. and Woolfson,** R.(1995a) *Prosecuting Child Abuse: An Evaluation of the Government's Speedy Progress Policy.* London: Blackstone Press.

Plotnikoff, J. and Woolfson, R. (1995b) Research Findings No. 29: The Child Witness Pack – An Evaluation. London: Home Office Research and Planning Unit.

**Plotnikoff, J. and Woolfson, R.** (1996) 'Evaluation of Witness Service support for child witnesses.' In Victim Support (Ed), *Children in Court*. London: Victim Support.

**Plotnikoff, J. and Woolfson, R.**(1998) *Policing Domestic Violence: Effective Organisational Structures*, Police Research Series Paper 100, London: Home Office Policing and Reducing Crime Unit.

**Poole, D. and Lamb, M.E.** (1998) *Investigative Interviews of Children: A Guide for Helping Professionals*. Washington: American Psychological Association.

**Poole, D. and Lindsay, D.S.** (1995) 'Interviewing preschoolers: Effects of nonsuggestive techniques, parental coaching and leading questions on reports of nonexperienced events.' *Journal of Experimental Child Psychology*, 129-154.

**Poole, D. and Lindsay, D.S.** (1998) 'Assessing the accuracy of young children's reports: Lessons from the investigation of child sexual abuse.' *Applied and Preventive Psychology*, 7, 1-26.

**Poole, D. and White, L.** (1991) 'Effects of question repetition on the eyewitness testimony of children and adults.' *Developmental Psychology*, 27, 975-986.

**Poole, D. and White, L.** (1993) 'Two years later: Effects of question repetition and retention interval on eyewitness testimony of children and adults.' *Developmental Psychology*, 29, 844-853.

**Porter, S., Yuille, J.C. and Bent, A.** (1995) 'A comparison of the eyewitness accounts of deaf and hearing children.' *Child Abuse and Neglect*, 19, 51-61.

**Raskin, D. and Esplin, P.** (1991) 'Statement validity assessment: Interview procedures and content analysis of children's statements of sexual abuse.' Behavioural Assessment, 13, 265-291.

Ricci, C.M., Beal, C.R. and Dekle, D.J. (1996) 'The effect of parent versus unfamiliar interviewers on children's eyewitness memory and identification accuracy'. *Law and Human Behavior*, 20, 483-500.

Roberts, H. and Glasgow, D. (1993) 'Gathering evidence from children: A systematic approach'. In N.K. Clark and G.M. Stephenson (Eds), *Children, Evidence and Procedure*. Leicester: British Psychological Society.

Roberts, K.P. and Lamb, M.E. (1999) 'Children's Responses When Interviewers Distort Details During Investigative Interviews', *Legal and Criminological Psychology*, 4, 23-32.

- Salmon, K., Bidrose, S. and Pipe, M-E. (1995) 'Providing props to facilitate children's event reports: A comparison of toys and real items.' *Journal of Experimental Child Psychology*, 60, 174-194.
- Sas, L.D., Cunningham, A.H., Hurley, P., Dick, T. and Farnsworth, A. (1995) *Tipping the Balance to Tell the Secret: Public Discovery of Child Sexual Abuse.* London, Ontario: London Family Court Clinic.
- Saunders, A., Creaton, J., Bird, S. and Weber, L . (1997) Victims with Learning Disabilities: Negotiating the Criminal Justice System. Oxford: University of Oxford Centre for Criminological Research.
- **Saywitz, K.** (1995) 'Improving children's testimony: The question, the answer and the environment.' In M. Zaragoza, G. Graham, G.N. Hall, R. Hirschman and Y. Ben-Porath (Eds), *Memory and Testimony in the Child Witness.* Thousand Oaks, CA: SAGE.
- **Saywitz, K. and Elliott, D.** (in press) *Interviewing Children in the Forensic Context:* A Developmental Approach. Washington, DC: American Psychological Association.
- **Saywitz, K., Geiselman, R. and Bornstein, G**. (1992) 'Effects of Cognitive Interviewing and Practice on Children's Recall Performance', *Journal of Applied Psychology*, 77: 744-756.
- **Saywitz, K. and Goodman, G.S**. (1993) 'Interviewing children in and out of court: Current research and practical implications.' In L. Berliner, J. Briere and J. Bulkley (Eds), *The APSAC Handbook on Child Maltreatment*. Thousand Oaks, CA: SAGE.
- Saywitz, K., Goodman, G.S., Nicholas, E. and Moan, S . (1991) 'Children's memories of a physical examination involving genital touch: Implications for reports of child sexual abuse.' *Journal of Consulting and Clinical Psychology*, 59, 682-691.
- Saywitz, K. and Lyon, T. (1997) 'Sensitively Assessing Children's Testimonial Competence.' Final report to the National Center on Child Abuse and Neglect, Grant No. 90-CA-1553. Los Angeles, CA: University of California, Los Angeles.
- **Saywitz, K., Nathanson, R. and Snyder, L** . (1993a) 'Credibility of child witnesses: The role of communicative competence.' *Topics in Language Disorders*, 13, 59-78.

- Saywitz, K., Nathanson, R., Snyder, L. and Lamphear, V . (1993b) Preparing Children for the Investigative and Judicial Process: Improving Communication, Memory and Emotional Resiliency. Final Report to the National Center on Child Abuse and Neglect (Grant No. 90CA1179)
- Saywitz, K. and Snyder, L. (1993) 'Improving children's testimony with preparation.' In G.S. Goodman and B. Bottoms (Eds), *Child Victims*, *Child Witnesses*: *Understanding and Improving Testimony*. New York: Guilford Press. Saywitz, K. and Snyder, L. (1996) 'Narrative elaboration: Test of a new procedure for questioning children.' *Journal of Consulting and Clinical Psychology*, 64, 1347-1357.
- **Saywitz, K., Snyder, L. and Lamphear, V**. (1996) 'Helping children tell what happened: A follow-up study of the narrative elaboration procedure.' *Child Maltreatment*, 1, 200-212.
- **Sigelman, C.K., Budd, E.C., Spanhel, C.L. and Schoenrock, C.J.** . (1981) 'Asking questions of retarded persons: A comparison of yes-no and either/or formats.' *Applied Research in Mental Retardation*, 347-357.
- **Simkins, L. and Renier, A.** (1996) 'An analytical review of the empirical literature on children's play with anatomically detailed dolls.' *Journal of Child Sexual Abuse*, 5, 21-45.
- **Sorenson, T. and Snow, B**. (1991) 'How children tell: The process of disclosure in child sexual abuse.' *Child Welfare*, 70, 3-15.
- **Sternberg, K.J., Lamb, M.E., Hershkowitz, I., Esplin, P** .**W. and Sunshine, N** . (1996) 'The relationship between investigative utterance types and the informativeness of child witnesses.' *Journal of Applied Developmental Psychology*, 17, 439-451.
- Steward, M., Bussey, K., Goodman, G. and Saywitz, K. (1993) 'Implications of developmental research for interviewing children.' *Child Abuse and Neglect*, 17, 25-37.
- **Stockdale, M. (1996)** An Evaluation of Interview Techniques used by Police Officers and Social Workers with Sexually Abused Children. Leicester University: Unpublished Masters Thesis.

**Thompson, R., Clark e-Stewart, A. and Lepore, S.** (1997) 'What did the janitor do? Suggestive interviewing and the accuracy of children's accounts.' *Law and Human Behavior*, 21, 405-426.

Utting, W., Baines, C., Stuart, M., Rowlands, J. and Vialva, R . (1997) People Like Us. The Report of the Safeguards for Children Living Away from Home. Department of Health, London: The Stationery Office.

Vandermaas, M., Hess, T., and Baker -Ward, L. (1993) 'Does anxiety affect children's memory for a stressful event?' Applied Cognitive Psychology, 7, 109-127.

**Vizard, E.** (1991) 'Interviewing children suspected of being sexually abused: A review of theory and practice.' In C.R. Hollin and K. Howells (Eds), *Clinical Approaches to Sex Offenders and their Victims*. Chichester: Wiley.

**Wade, A.** (1997) The Child Witness and the Criminal Justice Process: A Case Study in Law Reform. Leeds University: Unpublished PhD thesis.

**Wade, A. and Westcott, H.L.** (1997) 'No easy answers: Children's perspectives on investigative interviews.' In H.L. Westcott and J. Jones (Eds), *Perspectives on the Memorandum. Policy, Practice and Research in Investigative Interviewing.* Aldershot, Hants: Arena.

Walker, A.G. (1993) 'Questioning young children in court: A linguistic case study'. Law and Human Behavior, 17, 59-81.

**Walker, A.G.** (1994) Handbook on Questioning Children: A Linguistic Perspective. Washington: ABA Center on Children and the Law.

Walker, N.E. and Hunt, J.S. (1998) 'Interviewing child victim-witnesses: How you ask is what you get.' In C.P. Thompson, D.J. Herrmann, J.D. Read, D. Bruce, D.G. Payne and M.P. Toglia (Eds), Eyewitness Memory: Theoretical and Applied Perspectives. Mahwah, NJ: Lawrence Erlbaum Associates.

Warren, A. and McGough, L. (1996) 'Research on children's suggestibility: Implications for the investigative interview'. *Criminal Justice and Behavior*, 32, 269-303.

Warren, A., Wooddall, C.E., Hunt, J.S. and Perry, N. (1996) "It Sounds Good in Theory But..." Do investigative interviewers follow guidelines based on memory research?", Child Maltreatment, 1, 231-245.

Welch-Ross, M., Diecidue, K. and Miller, S . (1997) 'Young children's understanding of conflicting mental representation predicts suggestibility.' *Developmental Psychology*, 33, 43-53.

**Westcott, H.L.** (1994) 'The Memorandum of Good Practice and children with disabilities.' *Journal of Law and Practice*, 3, 21-32.

**Westcott, H.L.** (1995) 'Children's experiences of being examined and cross-examined: The opportunity to be heard?' *Expert Evidence*, 4, 13-19.

**Westcott, H.L. and Cross, M.** (1996) This Far and No Further: Towards Ending the Abuse of Disabled Children. Birmingham: Venture Press.

**Westcott, H.L. and Davies, G.M.** (1996a) 'Memorandum training in ACPCs: A survey.' *Journal of Practice and Staff Development*, 5, 48-64.

Westcott, H.L. and Davies, G.M. (1996b) 'Sexually abused children's and young people's perspectives on investigative interviews.' *British Journal of Social Work*, 26, 451-474.

Westcott, H.L., Davies, G.M. and Horan, N  $\,$ . (1998) Evaluating investigative interviews for suspected sexual abuse carried out by English police officers. Paper presented to the BPS Developmental Section Annual Conference, Lancaster, September 11-14.

**Westcott, H.L. and J. Jones** (1997) (Eds), *Perspectives on the Memorandum. Policy, Practice and Research in Investigative Interviewing*. Aldershot, Hants: Arena.

Westcott, H.L. and Page, M. (1998) Worse than being abused: Cross-examination and child witness identity. Paper presented to the BPS Social Section Annual Conference, Kent, September 21-23.

Whitcomb, D., Runyan, D.K., De Vos, E., Hunter, W.M., Cross, T.P., Everson, M.D., Peeler, N.A., Porter, C.Q., Toth, P. A. and Cropper, C. (1991) Child Victim as Witness Research and Development Program. Final Report to the Office of Juvenile Justice and Delinquency Prevention, US Department of Justice (Grant No. 87-MC-CX-0026).

**Wilson, J.C.** (1995) 'Developmental issues concerning testing children's evidence in court.' Expert Evidence, 4, 20-24.

**Wilson, C. and Pipe, M-E.** (1989) 'The effects of cues on young children's recall of real events.' *New Zealand Journal of Psychology*.

Wood, J.M., McClure, K.A. and Birch, R.A. (1996) 'Suggestions for improving interviews in child protection agencies.' *Child Maltreatment*, 1, 223-230.

Yamamoto, K., Davis, O.L., Dylak, S., Whittaker, J., Marsh, C. and van der Westhuizen, P.C. (1996) 'Across six nations: Stressful events in the lives of children.' Child Psychiatry and Human Development, 26, 139-150.

# RECENT POLICE RESEARCH GROUP AND POLICING AND REDUCING CRIME UNIT PUBLICATIONS:

# **Policing and Reducing Crime Unit**

**Police Research Series papers** 

- Sex Offending Against Children: Understanding the Risk. Don Grubin. 1998.
- 100. **Policing Domestic Violence: Effective Organisational Structures.**Joyce Plotnikoff and Richard Woolfson. 1999.
- Pulling the Plug on Computer Theft. Paula Whitehead and Paul Gray. 1999.
- 102. Face Value? Evaluating the Accuracy of Eyewitness Information.
  Mark R. Kebbell and Graham F. Wagstaff. 1999.
- 103. **Applying Economic Evaluation to Policing Activity.** J.E. Stockdale, C.M.E. Whitehead and P.J.Gresham. 1999.
- 104. Arresting Evidence: Domestic Violence and Repeat Victimisation. Jalna Hanmer, Sue Griffiths and David Jerwood. 1999.
- Proactive Policing on Merseyside. Alana Barton and Roger Evans.
   1999.
- 106. Tenure: Policy and Practice. Gary Mundy. 1999.
- 107. Career Progression of Ethnic Minority Police Officers. Nick Bland, Gary Mundy, Jacqueline Russell and Rachel Tuffin. 1999.
- 108. Preventing Residential Burglary in Cambridge: From Crime Audits to Targeted Strategies. Trevor Bennett and Linda Durie. 1999.
- 109. Policing Drug Hot-Spots. Jessica Jacobson. 1999.
- 110. Understanding and preventing police corruption: Lessons from the Literature. Tim Newburn. 1999.
- 111. On a Course: Reducing the Impact of Police Training on Availability for Ordinary Duty. Peter R. Langmead Jones. 1999.
- 112. Hot Products: Understanding, Anticipating and Reducing Demand for Stolen Goods. Ronald V. Clarke. 1999.
- 113. Consolidating Police Crackdowns: Findings from an Anti-Burglary Project. Graham Farrell, Sylvia Chenery and Ken Pease. 1999.
- 114. **Missing Presumed: The Police Response to Missing Persons.** Geoff Newiss. 1999.